



Registration of a Charge

Company Name: **TAG AVIATION (STANSTED) LTD** Company Number: **06137634**

Received for filing in Electronic Format on the: **25/06/2021**

Details of Charge

- Date of creation: 16/06/2021
- Charge code: 0613 7634 0004
- Persons entitled: LOMBARD NORTH CENTRAL PLC
- Brief description: AIRCRAFT BOEING 737-700 NATIONALITY BRITISH REG MARK G-NEWG SERIAL NO: 29885 FOR FULL DESCRIPTION PLEASE REFER TO THE INSTRUMENT
 - Contains fixed charge(s).
 - Contains negative pledge.

Authentication of Form

This form was authorised by: a person with an interest in the registration of the charge.

Authentication of Instrument

Certification statement: I CERTIFY THAT SAVE FOR MATERIAL REDACTED PURSUANT TO S.859G OF THE COMPANIES ACT 2006 THE ELECTRONIC COPY INSTRUMENT DELIVERED AS PART OF THIS APPLICATION FOR REGISTRATION IS A CORRECT COPY OF THE ORIGINAL INSTRUMENT.

Certified by: MARIE SLINN



XA7D7BL5

06137634



CERTIFICATE OF THE REGISTRATION OF A CHARGE

Company number: 6137634

Charge code: 0613 7634 0004

The Registrar of Companies for England and Wales hereby certifies that a charge dated 16th June 2021 and created by TAG AVIATION (STANSTED) LTD was delivered pursuant to Chapter A1 Part 25 of the Companies Act 2006 on 25th June 2021.

Given at Companies House, Cardiff on 28th June 2021

The above information was communicated by electronic means and authenticated by the Registrar of Companies under section 1115 of the Companies Act 2006





Aircraft Mortga	ge	· · · · · · · · · · · · · · · · · · ·	Lombard
ENGLAND OR IN ENG	NT DEED AND SHOULD ONLY BE USE GLISH AIRSPACE AT THE TIME THIS DE ITUS IS OBTAINED. YOU SHOULD TAKE	ED IS CREATED UNLESS	FOREIGN LEGAL ADVICE
Date	6 June 2021	(to be completed by Lombar	 ۱۹۹۵ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ ۱۹۹۵ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰
Parties			
Lombard North Central whose address for notices 8778888 Text Relay: 18001 0345 8	PLC, a company registered in England and Wal s is PO Box 520, Rotherham, S63 3BR. Email: ç 177 8888	es (registered number: 0033700 generalenquiries@lombard.co.ul	i4) «. Customer service: 0345
Outomode news			
Customer's name	TAG Aviation (Stansted) Ltd		
Customer's address	The Diamond Hangar, Long Border Road	, Stansted Airport, Essex, Eng	land, CM24 1RE
Customer's Jurisdiction	of Incorporation England and Wales		
Company / LLP Registra	tion number 06137634		
Address for Service (must be in the United Kingdom and include name)	The Diamond Hangar, Long Border Road,	Stansted Airport, Essex, Eng	land, CM24 1RE
Telephone Number:	01279 815544	Fax Number: N	
Contact:	Mr Trevor Gunn	 Construction Construction Construction Construction 	
Is the Customer a SPV?	No		
Operator			an general Street, et earlier and ender a street and a st
Name:	N/A	Company / LLP Registrat number:	on N/A
Operator address:	N/A		
1	************************************	7777477425400000000000000000000000000000	
	n an an Araban ann an Araban a Araban an Araban an Ar		م المحافظ المح محافظ المحافظ ال

	Original	Copy -	Page 2	2 of 2
--	----------	--------	--------	--------

Aircraft Details		
Registration Mark	G-NEWG	- .
Aircraft Manufacturer	The Boeing Company	
Aircraft Type	Boeing 737-700	
Year of Manufacture	2002	
Manufacturer's Serial Number	29885	

	Quantity	Manufacturer / Type	Serial Numbers
Engines	2	CFM / CFM56 -7B22	888920 & 888940
Avionics			N
Propeller / Rotor blades			
APU			

			<u>A second s second second sec second second sec</u>
State of Registration	tate of Registration United Kingdom		
Aviation Authority	UK Civil Aviation Authority		
Maintenance Performer	N/A		
Habitual Base	Lasham Airfield		999 Anni
			DEEX COLOR DE 2010 COLOR A
Required Insurance Value	The greater of 110% of the balance outstanding or the current market value of the aircraft	Required Insurance Currency	Sterling
Minimum Liability Cover	£322,000,000.00	Permitted Deductible	£10,000.00
Major Damage Sum	£10,000.00		N

Cape Town

Each Engine is 550 or more rated takeoff horsepower or its equivalent	Yes		
Each Propeller is capable of absorbing 750 or more related takeoff shaft	horsepow		

The Airframe is described on the International Registry Manufacturer's List as BOEING model 737-700 with serial number 29885

The Engine(s) is/are described on the International Registry Manufacturer's List as CFM model CFM56-7 with serial number 888920 & 888940

LAFG/MAA/---/1018 (T)

1	Definitions and interpretation	
1.1	In this Deed the following words will have the	e meanings given alongside them:
1.1.1	Aircraft	means the aircraft more particularly described in the Aircraft details
		together with the Engines and Propellers (whether or not any of the Engines and Propellers may from time to time be installed on the Airframe), the Technical Records and all Parts now or hereafter installed in or on the Airframe and all substitutions, renewals and replacements from time to time made in or to or installed in or on the Airframe in accordance with this Deed including any Parts which are for the time being detached from the Airframe but which remain the property of the Customer;
1.1.2	Airframe	means the Aircraft excluding the Engines and the Technical Records;
1.1.3	Airworthiness Directive	means an airworthiness directive issued by the Aviation Authority or by EASA or the national civil aviation authority of the Aircraft 's country of manufacture;
1.1.4	Authorisation	has the meaning given to it in the Aviation Loan Terms;
1.1.5	Authority Letter	the letters sent or to be sent by the Customer to any party (including Eurocontrol) in respect of any charges (including navigation charges) in relation to the Aircraft in the form set out in schedule 5;
1.1.6	Aviation Authority	each person who shall from time to time be vested with the control and supervision of, or have jurisdiction over, the registration, airworthiness and operation of aircraft or other matters relating to civil aviation in the State of Registration;
117	Aviation Loan Agreement	the aviation loan agreement entered into between the Customer and Lombard in relation to the financing of the Aircraft inclusive of the Aviation Loan Terms;
1.1.8	Avlation Loan Terms	the aviation loan terms which form part of the Aviation Loan Agreement;
1,1.9	Business Day	has the meaning given to it in the Aviation Loan Terms;
1.1.10	Cape Town Convention and Aircraft Protocol	means the Convention on International Interests in Mobile Equipment and the Aircraft Protocol on International Interests in Mobile Equipment on Matters specific to Aircraft Equipment each signed at Cape Town on the 16th November, 2001;
1.1.11	Charging Period	has the meaning given to it in the Aviation Loan Terms;
1.1.12	Charter	any charter relating to the Aircraft or any other contract for use or employment of the Aircraft pursuant to clause 9;
1.1.13	Charter Guarantee	any guarantee, bond, letter of credit or any other form of security supporting a Charter;
1.1.14	Compulsory Acquisition	means, in respect of the Aircraft, the Airframe and/or an Engine, requisition for title or other compulsory acquisition of title (but excluding requisition for use or hire) of such Aircraft, Airframe or Engine, as the case may be;
1 1 15	Currency	has the meaning given to it in the Aviation Loan Agreement;
1.1.16	Customer's Obligations	has the meaning given to it at clause 2;
1.1.17	Default	has the meaning given to it in the Aviation Loan Terms;
1.1.18	Deregistration Power of Attorney	a power of attorney to deregister the Aircraft in favour of Lombard in the form set out in Schedule 1;

1,1.19	EASA	the European Aviation Safety Authority and any successor organisation
		authority;
1.1.20	Earnings	all amounts of any kind which may at any time be earned, due or payab arising from the use, ownership, possession or operation of the Aircraft ;
1.1.21	Encumbrance	has the meaning given to it in the Aviation Loan Terms;
1.1.22	Engines	 (a) each of the engines described in the Aircraft details whether or n from time to time installed on the Airframe or any other airfram but which, having been removed from the Airframe, remains th property of the Customer;
		(b) any other engine which may from time to time be installed upon attached to the Airframe which becomes the property of the Customer ;
		(c) in so far as the same belong to the Customer , any and a appliances, instruments or accessories or other equipment or Part of whatever nature from time to time relating to an engine referre to in (a) or (b) above whether or not installed on or attached to suc engine; and
		(d) in so far as the same belongs to the Customer , all substitution replacements or renewals from time to time made on or to ar items referred to in (a), (b) and (c) above in accordance with th Deed ;
1.1.23	Eurocontrol	the Central Route Charges Office, European Organisation for the Safety Air Navigation;
1.1.24	Eurocontrol Letter	means the letter sent or to be sent by the Customer to Eurocontrol respect of air navigation charges in relation to the Aircraft in the form s out in Schedule 4;
1.1.25	Event of Default	has the meaning given to it in the Aviation Loan Terms;
1:1.26	Finance Document	has the meaning given to it in the Aviation Loan Terms;
1.1.27	Financial Indebtedness	has the meaning given to it in the Aviation Loan Terms;
1.1.28	Guarantor	has the meaning given to it in the Aviation Loan Terms;
1.1.29	Holding Company	has the meaning given to it in the Aviation Loan Terms;
1.1.30	Insurance(s)	all policies or contracts of insurance relating to the Aircraft;
1.1.31	Insurers	in respect of the Insurances , such insurance underwriters and/or insurance companies in the major international insurance markets as may be, fro time to time, approved by Lombard in accordance with the terms of the Deed ;
1.1.32	Interest	has the meaning given to it in clause 2.1.1;
1.1.33	International Registry	the registry established pursuant to the Cape Town Convention ar Aviation Protocol;
1.1.34	Maintenance Agreement	has the meaning given to it in the Aviation Loan Terms;
1.1.35	Maintenance Performer	has the meaning given to it in the Aviation Loan Terms;
1.1.36	Maintenance Programme	a programme for the maintenance of the Aircraft prepared and/or approve by the Aviation Authority and the manufacturer of the Aircraft;
1.1.37	Major Damage	has the meaning given to it in clause 5.2.2;

1.1.39 Material Adverse Effect a change in condition or discussion activation of a single adverse (adverse) begin any Obligor sum of its psyment or other obligations under the under the Finance Documents or any other agreement with Lombard; 1.140 Obligor the Customer, any Guarantor and any Security Provider; 1.141 Operating Agreement has the meaning given to it in the Avlation Loan Terms; 1.142 Operator such person operating, chartering or leasing the Alvereft as parmitted under daues 9; 1.143 Parts means, all modules, applances, parts, accessries, auxiliary power unit, instanting, furnishings and other explorent of whatsoever nature which are formiting, furnishings and other explorent of whatsoever nature which are formiting, furnishings and other explorent of whatsoever nature which are formiting, furnishings and other explorent of whatsoever nature which are formiting, furnishings and other explorent of whatsoever nature which are formiting, furnishings and other explorent of whatsoever nature which are formiting, furnishings and other explorent of whatsoever nature which are formiting, furnishings and other explorent of whatsoever nature which are formiting, furnishing and the explorent of whatsoever nature which are formiting in the Alveraft details whether or not form time to its the Alveraft details whether or not form time to its the Alveraft details whether or not form time to its in the Alveraft details whether or not form time to its in the Alveraft details whether or not form time to the restant on any other propeller which may from time to time and the customer; and or on other exploteneme or any other ar	1.1.38	Major Damage Sum	the r	najor damage sum specified at the front of this Deed;
1.140 Obligor the Customer, any Guarantor and any Security Provider: 1.141 Operating Agreement has the meaning given to il in the Aviation Loan Terms; 1.142 Operator such person operating, chartering or leasing the Alerant as permitted under clause 9; 1.143 Parts means, all modules, appliances, parts, accessories, auxiliary power unit, instruments, turnishings and other equipment of whatsoever nature which are from time to time attached to the Alframe or an Engline or which, having been removed thereform, remains the property of the Customer; 1.143 Parts means, all modules, appliances, parts, accessories, auxiliary power unit, instruments, turnishings and other equipment of whatsoever nature which are from time to time to tare of the Customer; 1.144 Permitted Lian has the meaning given to it in the Aviation Loan Terms; 1.145 Permitted Transoction has the meaning given to it in the Aviation Loan Terms; 1.146 Propellers means. (a) each of the propellers described in the Alframe or any other alframe to third the property of the Customer, and (b) any other propeller which may from time to time be installed upon to a statched to the Alframe or any other alframe, remeins the property of the Customer, and (c) in so far as the same belong to the Customer, all substitutions, replacements for mine to time to time relating to a propelier referred to in (a) to to above thether or	1.1.39	Material Adverse Effect	treat to th Obli to m	ment) which in Lombard's view is materially adverse to any Obligor or e assets, business or financial condition or trading prospects of any gor such that the Obligor could reasonably be expected to be unlikely set any of its payment or other obligations under the under the Finance
11.42 Operator such person operating, chartering or leasing the Aircraft as permitted under clause 9; 11.43 Parts means, all modules, appliances, parts, accessories, auxiliary power unit, instruments, fumishings and other aupinent of whatsoever nature which, aving been removed therefrom, remains the property of the Customer, 11.44 Permitted Lien has the meaning given to it in the Aviation Loan Terms; 11.45 Permitted Transaction has the meaning given to it in the Aviation Loan Terms; 11.46 Propellers means: (a) each of the propellers described in the Alricraft details whether or not from time to time installed on the Alriframe, remains the property of the Customer, or (b) any other propeller which may from time to time be installed upon or attached to the Alriframe, remains the property of the Customer, or (c) in so far as the same belong to the Customer, any and all appliances, instruments or accessories or other equipment of Parts of whatever nature from time to time relating to a propeiler of farts of whatever nature which becomes with the Deed; 1.1.47 Relevant Jurisdiction has the meaning given to it in the Aviation Loan Terms; 1.1.48 Requisition (a) any other propeller which may from time to time relating to a propeller of farts of whatever nature which becomes the property of the Customer; and 1.1.47 Relevant Jurisdiction has the same belongs to the Customer; ano	1.1.40	Obligor		
 clause 9: 11.43 Parts means, all modules, appliances, parts, accessories, auxiliary power unit, instruments, funitatings and other ecuipment of whatsoever nature which are from time to time attached to the Airframe or an Engine or which, having been removed therefrom, remains the property of the Customer, 11.44 Permitted Lien has the meaning given to it in the Aviation Loan Terms; 11.45 Permitted Transaction has the meaning given to it in the Aviation Loan Terms; 11.46 Propeilers (a) each of the propeilers described in the Aircraft details whether or not from time to time installed on the Airframe or any other airframe but which, having been removed from the Airframe, remains the property of the Customer; or (b) any other propailer which may from time to time be installed upon or attached to the Airframe witch becomes the property of the Customer; and (c) in so far as the same belong to the Customer, any and all appliances, instruments or accessories or other equipment of Parts of whatever nature from time to time to time relating to a propeiler referred to in (a) (b) above whether or no installed on or tatached to such engine; and in so far as the same belongs to the Customer, all substitutions, replacements or meawails from time to time made on or to any items referred to in (a) (b) above whether or not installed to any detain referred to in (a) (b) above whether or nor installed on or acquisition of the Aircraft whether for full consideration accusteriation any tems referred to in (a) (b) above whether or no installed on or acquisition of the Aircraft whether for full consideration activation (a) (b) above whether or no installed on or acquisition of the Aircraft whether for full consideration or explores referred to in (a) (b) above whether or no installed on or acquisition of the Aircraft in the Aircraft (including any higaching or onsideration consider	1.1.41	Operating Agreement	has	the meaning given to it in the Aviation Loan Terms;
instruments. furnishings and other squipment of whatsoever nature which are from time to time attached to the Alrframe or an Engine or which, having been removed therefrom, mennas the property of the Customer; 1.1.44 Permitted Lien has the meaning given to it in the Aviation Loan Terms; 1.1.45 Permitted Transaction has the meaning given to it in the Aviation Loan Terms; 1.1.46 Propellers means. (a) each of the propellers described in the Alrframe or any other airframe but which, having been removed from the Alrframe, remains the property of the Customer; or (b) any other propeller which may from time to time be installed upon or attached to the Alrframe which becomes the property of the Customer; and (c) in so far as the same belong to the Customer, any and all appliances, instruments or accessories or other equipment of whatsoever nature from time to time nation to any items referred to in (a) (b) and (c) above whether or not installed on or attached to such engine; and 1.1.47 Relevant Jurisdiction in so far as the same belongs to the Customer, all substitutions, replacements or renewals from time to time made on or to any items referred to in (a). (b) and (c) above whether or not installed on or whoth any oronideration, a consideration is ensultion be arbitration which is effected by any government or official substitutions, includeration, which is effected by any government or official substitution (a) (c) any ary person or persons claiming to be or to represent to from time to time nation in the alreraft (including ane yhisticin or a fixed period not exceeding one year without any	1,1.42	Operator		
1.145 Permitted Transaction has the meaning given to it in the Aviation Loan Terms; 1.1.46 Propellers means: (a) each of the propellers described in the Aircraft details whether or not from time to time installed on the Airframe or any other airframe but which, having been removed from the Airframe, remains the property of the Customer; or (b) any other propeller which may from time to time be installed upon or attached to the Airframe which becomes the property of the Customer; and (c) in so far as the same belong to the Customer, any and all appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time relating to a propeller referred to in (a) or (b) above whether or not installed on or attached to such engine; and 1.1.47 Relevant Jurisdiction 1.1.48 Requisition (a) any expropriation, confiscation, requisition or acquisition of the Aircraft, whether for till consideration, a consideration is through the air of any items referred to in (a). (b) above whether or not installed on or a capusition of the Aircraft, whether for till consideration a consideration is the proper value, a nominal consideration is the or or present or person or person or person or the represent or official authority (or by any person or person discration is the or person a section by any consideration, which is effected by any government or official authority (excluding a requisition for hire for a fixed period not exceeding one year without any right to an extension); and / or (b) any arrest, capture, seizure, sequestration, foreliture	1,1.43	Parts	instru are f	uments, furnishings and other equipment of whatsoever nature which rom time to time attached to the Airframe or an Engine or which, having
1.1.46 Propellers means: (a) each of the propellers described in the Aircraft details whether or not from time to time installed on the Airframe or any other airframe but which, having been removed from the Airframe, remains the property of the Customer; or (b) any other propeller which may from time to time be installed upon or attached to the Airframe which becomes the property of the Customer; and (c) in so far as the same belong to the Customer, any and all appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time made on or to any items referred to in (a) or (b) above whether or not installed on or attached to such engine; and 1.1.47 Relevant Jurisdiction 1.1.48 Requisition (a) any expropriation, confiscation, requisition or accusition of the Aircraft, whether for full consideration are within a consideration ites than its proper value, a nominal consideration or without any consideration, whether for full consideration, and consideration or of the for autority devoluting a requisition for thre for a fixed period not exceeding one year without any right to an extension); and / or (b) any arrest, capture, seizure, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft is redelivered to the full control of the relevant Obligor within thirty (20) days:	1.1.44	Permitted Lien	has	he meaning given to it in the Aviation Loan Terms;
 (a) each of the propellers described in the Aircraft details whether or not from time to time installed on the Airframe or any other airframe but which, having been removed from the Airframe, remains the property of the Customer; or (b) any other propeller which may from time to time be installed upon or attached to the Airframe which becomes the property of the Customer; and (c) in so far as the same belong to the Customer, any and all appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time relating to a propeller referred to in (a) or (b) above whether or not installed on or attached to two the event of the customer, and (c) in so far as the same belongs to the Customer, all substitutions, replacements or renewals from time to time made on or to any items referred to in (a), (b) and (c) above in accordance with this Deed; 1.1.47 Relevant Jurisdiction 1.1.48 Requisition (a) any expropriation, confiscation, requisition or acquisition of the Aircraft, whether for full consideration, a consideration fees than tis proper value, a nominal consideration or official authority or by any person persons claiming to be or to represent a government or official authority or by any person or persons claiming to tany the for a fixed period not exceeding one year without any injut to an extension); and / or (b) any arrest, capture, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft is redelivered to the full control of the relevant Obligor within thirty (30) days; 1.1.49 Requisition Compensation 	etti. Setta			
or attached to the Airframe which becomes the property of the Customer; and (c) in so far as the same belong to the Customer, any and all appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time relating to a propeller referred to in (a) or (b) above whether or not installed on or attached to such engine; and 11.147 Relevant Jurisdiction 11.148 Requisition (a) any expropriation, confiscation, requisition or acquisition of the Aircraft, whether for full consideration, a consideration less than its proper value, a nominal consideration or without any consideration, which is effected by any government or official authority of yany person corperson claiming to be or to represent a government or official authority (excluding a requisition for hire for a fixed period not exceeding one year without any right to an extension); and / or (b) any arrest, capture, seizure, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft (including any hijacking or theft or disappearance). 11.149 Requisition Compensation				each of the propellers described in the Aircraft details whether or not from time to time installed on the Airframe or any other airframe but which, having been removed from the Airframe, remains the
appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time relating to a propeller referred to in (a) or (b) above whether or not installed on or attached to such engine; and in so far as the same belongs to the Customer, all substitutions, replacements or renewals from time to time made on or to any items referred to in (a), (b) and (c) above in accordance with this Deed; 1.1.47 Relevant Jurisdiction has the meaning given to it in the Aviation Loan Terms; 1.1.48 Requisition (a) any expropriation, confiscation, requisition or acquisition of the Aircraft, whether for full consideration, a consideration less than its proper value, a nominal consideration or without any consideration, which is effected by any government or official authority (excluding a requisition for the for a fixed period not exceeding one year without any right to an extension); and / or (b) any arrest, capture, seizure, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft (including any hijacking or theft or disappearance), in each case unless the Aircraft is redelivered to the full control of the relevant Obligor within thirty (30) days;			(b)	or attached to the Airframe which becomes the property of the
1.1.47 Relevant Jurisdiction has the meaning given to it in the Aviation Loan Terms; 1.1.48 Requisition (a) any expropriation, confiscation, requisition or acquisition of the Alrcraft, whether for full consideration, a consideration less than its proper value, a nominal consideration or without any consideration, which is effected by any government or official authority or by any person or persons claiming to be or to represent a government or official authority (excluding a requisition for hire for a fixed period not exceeding one year without any right to an extension); and / or (b) any arrest, capture, seizure, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft is redelivered to the full control of the relevant Obligor within thirty (30) days; 1.1.49 Requisition Compensation			(c)	appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time relating to a propeller referred to in (a) or (b) above whether or not installed on or attached to such
1.1.48 Requisition (a) any expropriation, confiscation, requisition or acquisition of the Aircraft, whether for full consideration, a consideration less than its proper value, a nominal consideration or without any consideration, which is effected by any government or official authority or by any person or persons claiming to be or to represent a government or official authority (excluding a requisition for hire for a fixed period not exceeding one year without any right to an extension); and / or (b) any arrest, capture, seizure, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft (including any hijacking or theft or disappearance), in each case unless the Aircraft is redelivered to the full control of the relevant Obligor within thirty (30) days; 1.1.49 Requisition Compensation			repla	cements or renewals from time to time made on or to any items referred
 Aircraft, whether for full consideration, a consideration less than its proper value, a nominal consideration or without any consideration, which is effected by any government or official authority or by any person or persons claiming to be or to represent a government or official authority (excluding a requisition for hire for a fixed period not exceeding one year without any right to an extension); and / or (b) any arrest, capture, seizure, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft (including any hijacking or theft or disappearance), in each case unless the Aircraft is redelivered to the full control of the relevant Obligor within thirty (30) days; 1.1.49 Requisition Compensation all sums payable by reason of a Requisition and/or Compulsory 	1.1.47	Relevant Jurisdiction	has t	he meaning given to it in the Aviation Loan Terms;
acquisition or detention of the Aircraft (including any hijacking or theft or disappearance), in each case unless the Aircraft is redelivered to the full control of the relevant Obligor within thirty (30) days; 1.1.49 Requisition Compensation all sums payable by reason of a Requisition and/or Compulsory	1.1.48	Requisition	(a)	Aircraft, whether for full consideration, a consideration less than its proper value, a nominal consideration or without any consideration, which is effected by any government or official authority or by any person or persons claiming to be or to represent a government or official authority (excluding a requisition for hire for a fixed period not exceeding one year without any right to an
relevant Obligor within thirty (30) days; 1.1.49 Requisition Compensation all sums payable by reason of a Requisition and/or Compulsory			(b)	acquisition or detention of the Aircraft (including any hijacking or
	1.1.49	Requisition Compensation		

ſ

A x		
1.1.50	Security Assets	the Aircraft and all assets which from time to time are subject to the security created or expressed to be created by this Deed ;
1.1.51	Security Documents	has the meaning given to it in the Aviation Loan Agreement;
1.1.52	Security Period	the period starting from the date of this Deed and ending on the date on which the Customer's Obligations have been unconditionally and irrevocably repaid and discharged in full and Lombard has no commitment to lend to the Customer ;
1.1.53	SPV	has the meaning given to it in the Aviation Loan Terms;
1.1.54	State of Registration	has the meaning given to it in the Aviation Loan Agreement;
1.1.55	Subsidiary	has the meaning given to it in the Aviation Loan Terms;
1.1.56	Tax	has the meaning given to it in the Aviation Loan Terms;
1.1.57	Technical Records	means all records, logs, technical data and other materials and documentation relating to the Aircraft and the maintenance and operation thereof, including, without limitation, those which are from time to time required to comply with the regulations and requirements of the State of Registration or which are kept in purported compliance with such regulations and requirements;
1.1.58	Total Loss	has the meaning given to it in the Aviation Loan Terms;
1.1.59	VAT	has the meaning given to it in the Aviation Loan Terms; and
1.1.60	Warranties	any agreement under which warranties are provided in favour of the Customer in relation to the Aircraft .
1.2 1.3	defined in this Deed .	Loan Agreement have the same meaning in this Deed unless expressly viation Loan Agreement apply to this Deed as though they were set out in
1.0	full in this Deed , except that references to the	e Aviation Loan Agreement apply to this Deed as though they were set out in the Aviation Loan Agreement will be construed as references to this Deed.
1.4	Unless a contrary indication appears a refere than one person, includes such person's join them together and separately;	ence in this Deed to "Customer's Obligations" where the Customer is more in this Deed to "Customer" are to nt, several and independent liabilities and references to "Customer" are to
1.5	It is intended that this document takes effect	as a deed notwithstanding that a party may only execute it under hand.
2	Customer's Obligations	
2.1	The Customer agrees to pay to Lombard or all the Customer's liabilities to Lombard (pr another) including (without limitation):	n demand all the Customer's Obligations. The Customer's Obligations are resent, future, actual or contingent and whether incurred alone or jointly with
2.1.1	interest at the rate charged by Lombard in a before and after demand or judgment or dec	ccordance with the terms of the Aviation Loan Agreement, calculated both ree on a daily basis ("Interest"); and
2.1.2	from the date of payment) in connection w	es, Lombard or a receiver incurs (on a full indemnity basis and with Interest ith the Aircraft or taking, perfecting, protecting or enforcing this Deed or Lombard will provide details of its costs and expenses to the Customer.
2.2	Interest due and unpaid will be compounded and payable.	I monthly on the days selected by Lombard but will remain immediately due
2.3	be made in full, without set-off or deduction required by law in which event the Custome	nents under this Deed is absolute and unconditional and each payment must b, or withholding for any Taxes , save for any withholding Taxes which are ar shail pay to Lombard such additional amount as is necessary to ensure a equal to the amount Lombard would have received in the absence of any ction or withholding.
3	Charge	الایا این این این این این این این این این این این

....

As a continuing security for the payment on demand of the Customer's Obligations the Customer hereby 3.1 unconditionally and irrevocably and with full title guarantee: 3.1.1 by way of first priority legal mortgage, mortgages its respective interest in the Aircraft to Lombard; 3.1.2 assigns to Lombard absolutely by way of security all of its right, title, benefit and interest, now or at any later time, in and to: (a) all Charters, Earnings and any Charter Guarantee; (b) all Insurances and proceeds of the Insurances; (c) all Warranties, Operating Agreements and Maintenance Agreements: (d) all Reguisition Compensation; and any sale and purchase agreement which the Customer is party to in relation to the Aircraft. (e) The assignment under clause 3.1.2 above is subject to a proviso for reassignment on irrevocable discharge in full of the 3.2 Customer's Obligations. Following the expiry of the Security Period Lombard shall, at the request and cost of the Customer promptly execute and deliver to the Customer a discharge of this Deed together with all further instruments and documents which the Customer may require for the purpose of discharging the security constituted by this Deed. If any property expressed to be assigned in this clause 3 cannot be assigned, the Customer shall hold that property on 3.3 trust for Lombard and the assignment shall take effect as an assignment of any damages, compensation, remuneration, profit, rent or income which the Customer may derive from that right or be entitled to in respect of that right. 3.4 Any Part or Engine which becomes the property of the Customer pursuant to clause 8.2.5 shall, without the need for any further act, become subject to this Deed and shall form part of the Security Assets. 4 **Perfection of security** 4.1 The Customer shall promptly on request by Lombard, give notice of assignment: 4.1:1 in form and substance satisfactory to Lombard, to each party with whom the Customer has entered into a Charter, Charter Guarantee, Operating Agreement or Maintenance Agreement or any other agreement under which Earnings are paid; 4.1.2 substantially in the form set out in Schedule 2, to each Insurer; 4.1.3 substantially in the form set out in Schedule 3, to each person who has given Warranties or entered into a Charter, Charter Guarantee, Operating Agreement or Maintenance Agreement in relation to the Aircraft, and shall use all reasonable endeavours to procure that each addressee of such notice will promptly provide an acknowledgement to Lombard in form and substance satisfactory to Lombard. 5 Receipts 5.1 The Earnings and any sums recoverable under any Charter Guarantee, Operating Agreement or Maintenance Agreement shall be payable to the Customer until Lombard directs to the contrary following the occurrence of an Event of Default, on and after which direction: 5.1.1 the Customer shall immediately pay such amounts to Lombard or as Lombard may direct, and Lombard may instruct all persons from whom those sums are due to do the same; and 5.1.2 any such sums then held by the Customer's brokers, bankers or other agents or representatives shall be deemed to have been received by, and to be held by them, on trust for Lombard. Before an Event of Default occurs, sums recoverable in respect of the Insurances shall be payable as follows: 5.2 5.2.1 any sum recoverable in respect of a Total Loss under the Insurances shall be paid to Lombard; any sum recoverable under the Insurances in respect of a casualty to the Aircraft in respect of which the claim or the 5.2.2 aggregate of the claims against all insurers (before adjustment for any relevant franchise or deductible) exceeds the Major Damage Sum ("Major Damage") shall be paid to Lombard save that:

	(a)	the sum received by Lombard shall be paid over to the Custo satisfactory to Lombard that all loss and damage resulting from the repaired and that liabilities connected with the casualty have been	he casualty has been properly made good and
	(b)	the insurers with whom the Insurances are effected may in the c written consent of Lombard , make payment to the repairers on a out;	
5.2.3	accepte become	n recoverable in respect of a Total Loss of an Engine shall be paired a replacement engine for the Engine which has suffered a To an Engine for the purposes of this Deed) the sum received by Lo eller of that replacement Engine ;	tal Loss (and such replacement engine has
5.2.4	any othe loss and	er sum recoverable under the insurances shall be paid to the Cus d fully repairing all damage in respect of which that insurance mone	tomer which shall apply it in making good the ey was received; and
5.2.5	any sun whom th	n recoverable under the Insurances against general third party li he Customer is liable or to the Customer in reimbursement of mor	iability shall be paid directly to the person to neys expended by it to discharge that liability.
5.3		Iter the occurrence of an Event of Default , any sums recoverable un use 5.2.5) shall be payable to Lombard .	nder the Insurances (other than those referred
5.4	On or af to Lomb	fter the occurrence of an Event of Default, any sums recoverable i bard.	in respect of the Warranties shall be payable
6	Repres	entations and warranties	
		المائنة المربق محمد المراجع من المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع	المراجع بي محمد من محمد من يونين محمد من المراجع محمد محمد محمد مراجع مراجع المراجع المراجع المراجع المراجع ال محمد المراجع ال
6.1	Ine Cus	stomer represents and warrants to Lombard on the date it signs the	his Deed that:
6.1.1	it is a lii Jurisdic	mited liability company or a limited liability partnership duly incorn ction of Incorporation and it has the power to carry on its busines	porated or established under the laws of the as and to own its property and other assets;
6.1.2	the oblig	gations expressed to be assumed by it in each Finance Documen orceable obligations;	t to which it is a party are legal, valid, binding
6.1.3	it has po	ower to execute and perform its obligations under those Finance D	Ocuments to which it is a party;
6.1.4	the exec	cution and performance by it of the Finance Documents to which i	it is a party does not:
	(a)	contravene any applicable law, order or regulation;	
			المراجع المراجع المراجع المراجع
	(b)	conflict with, or result in any breach of any of the terms of, or cons instrument to which it is a party or any licence or other authorisat of its property is bound unless such conflict, breach or default of Material Adverse Effect ; or	tion to which it is subject or by which it or any
	(c)	contravene or conflict with its memorandum, articles of associated relating to it;	ation or any other constitutional documents
6.1.5	it has an	nd will maintain all necessary Authorisations to perform its obligat	tions under the Finance Documents;
6.1.6		thorisation required or desirable to make any Finance Docume ction has been obtained or effected and is in full force and effect;	ent admissible in evidence in each Relevant
6.1.7	the choi Relevan	ice of governing law of any Finance Document to which it is a p nt Jurisdiction ;	arty will be recognised and enforced in each
6.1.8	any judg of that F	gment obtained in relation to any Finance Document to which it is Finance Document will be recognised and enforced in each Releva	a party in the jurisdiction of the governing law ant Jurisdiction;
6.1.9	any filin necessa	g, recording or enrolment with any court or other authority of an ary under the laws of any Relevant Jurisdiction has been complet	ny Finance Document to which it is a party ted;
6.1.10	any stan	np, registration, notarial or similar Tax or fee to be paid (on or in re	elation to):
		ing, region anon, notation of annual rac of ree to be paid (off of infe	

- (b) the transactions contemplated by any Finance Document to which it is a party; or
- (c) under the laws of any Relevant Jurisdiction,
- has been paid or will be paid by the time required by law;
- 6.1.11 it is not necessary under the laws of any Relevant Jurisdiction:
 - (a) in order to enable Lombard to enforce its rights under any Finance Document to which the Customer is a party; or
 - (b) by reason of the execution of any **Finance Document** to which the **Customer** is a party or the performance by **Lombard** of its obligations under any **Finance Document** to which the **Customer** is a party,
 - that Lombard should be licensed, qualified or otherwise entitled to carry on business in any Relevant Jurisdiction;
- 6.1.12 Lombard will not be deemed to be resident, domiciled or carrying on business in any Relevant Jurisdiction by reason only of the execution, performance and/or enforcement of any Finance Document to which the Customer is a party;
- 6.1.13 it has not taken any action nor have any steps been taken or legal proceedings been started or to the best of its knowledge threatened in writing against it for:
 - (a) winding-up, dissolution, re-organisation, voluntary arrangement under Part I of the Insolvency Act 1986 or any other scheme of arrangement, restructuring plan, the granting of a moratorium or a compromise or composition with any of its creditors; or
 - (b) the enforcement of any Encumbrance over its assets; or
 - (c) the appointment of a receiver, administrative receiver, administrator, trustee or similar officer of it or of any of its assets;
- 6.1.14 except as disclosed to Lombard in writing before the date of signing this Deed, no action, litigation, arbitration or administrative proceeding has been commenced or to the best of its knowledge is pending or threatened against it or any member of the Customer (where the Customer is a LLP) or any Subsidiary of the Customer, involving a potential liability which could if adversely determined, individually or in aggregate, have a Material Adverse Effect nor is there subsisting any unsatisfied judgment or award given against it or any such member or Subsidiary;

6.1.15 except as disclosed to Lombard in writing before the date of signing this Deed, to the best of its knowledge, having made reasonable enquiry, no action, litigation, arbitration or administrative proceeding has been commenced or is pending or threatened against any Obligor (other than the Customer) or any Holding Company or member of such Obligor or Holding Company (where such Obligor or Holding Company is a LLP) involving a potential liability which could, individually or in aggregate, have a Material Adverse Effect nor is there subsisting any unsatisfied judgment or award given against any such Obligor, Holding Company or member;

6.1.16 it has not made any representations or warranties, whether express or implied, in connection with the Finance Documents which may increase Lombard's obligations or adversely affect Lombard's rights;

- 6.1.17 all information supplied by it or by any person on its behalf to Lombard was and remains true and complete in all respects and it is not aware of any material fact or circumstance which has not been disclosed to Lombard;
- 6.1.18 the Aircraft is not subject to any Encumbrance other than any Permitted Lien;
- 6.1.19 where it is stated or shown in the Deed that the Customer is a SPV:
 - (a) that no Encumbrance exists over all or any of its assets other than any Permitted Lien; and
 - (b) it has no Financial Indebtedness outstanding other than a Permitted Transaction;
- 6.1.20 the security created by each Security Document has or will have the ranking which it is expressed to have in the Security Document and it is not subject to any prior ranking or pari passu ranking Encumbrance other than as permitted in writing by Lombard or as exists by operation of applicable law;
- 6.1.21 it has a good, valid and marketable title to, or valid leases or licences of, and all appropriate Authorisations to use, the assets necessary to carry on its business as presently conducted;
- 6.1.22 It is the sole legal and beneficial owner of the assets over which it purports to grant security pursuant to the Security Documents to which it is a party including, without limitation, the Aircraft; and

6.1.23	where it is stated in this Deed, that the Aircraft shall be used in the course of the Customer's business only and not otherwise.
6.2	Each of the representations set out above are repeated by the Customer on the date of this Deed and the last day of each Charging Period by reference to the facts and circumstances then existing.
7	Aircraft undertakings
7.1	The Customer will at all times during the Security Period (or, if applicable, shall procure that any Operator of the Aircraft will):
7.1.1	register and keep the Aircraft registered at the Aviation Authority with the same registration mark either:
	(a) in the Customer's name as owner; or
	(b) if applicable, in the name of the Operator,
	in accordance with all applicable laws and regulations;
7.1.2	register this Deed and keep this Deed registered against the Aircraft in the register of aircraft mortgages maintained by the Aviation Authority ;
7.1.3	not do, and shall procure that no third party does, any act or thing which might prejudice or cancel the registrations referred to in clauses 7.1.1 and 7.1.2;
7.1.4	inform Lombard immediately of any requirement imposed by any insurer of the Aircraft or an authority with which the Customer is unable to comply;
7.1.5	use the Aircraft in a safe, careful and proper manner and in accordance with the manufacturer's recommendations;
7.1.6	provide to Lombard annually copies of any maintenance records, log books and insurance information relating to the Aircraft;
7.1.7	ensure the Aircraft is operated in accordance with all applicable laws and regulations (including those of the Aviation Authority) and in a manner which does not prejudice the rights of Lombard in respect of the Security Assets, not engage in any unlawful trade or allow the Aircraft to carry a cargo which will expose the Aircraft to penalty or forfeiture;
7.1.8	pay when due and discharge or cause to be promptly paid and discharged as they fall due all costs and expenses of operating, storing and maintaining the Aircraft and performing its obligations under the documents in relation to the Aircraft (including all taxes, fines, licence fees, charges of airport authorities (whether relating to landing fees, parking fees, handling charges or otherwise), all charges imposed by air navigation authorities (including Eurocontrol) and all charges of the Aviation Authority (whether relating to navigation or otherwise)) and will on written request by Lombard , supply evidence of the payment of any such charges;
7.1.9	immediately inform Lombard of any arrest, proceedings, collision, incident, or damage (where such damage exceeds the Major Damage Sum) or Total Loss affecting the Aircraft;
7.1.10	not, without the prior written consent of Lombard , change the Habitual Base from that notified to Lombard prior to the date of this Deed . This clause shall not apply where the Habitual Base is changed to a location in England provided always Lombard is given prior written notice of such location;
7.1.11	permit Lombard and its authorised representatives to inspect the Aircraft at any time upon giving reasonable notice and for that purpose grant or use all reasonable endeavours to procure access for Lombard and its authorised representatives to any land or buildings on or in which the Aircraft or any Engine or Part may be stored provided however that any such inspection shall not (unless an Event of Default or a Default has occurred) disturb the normal commercial operation or maintenance of the Aircraft;
7.1.12	not allow the Aircraft to be used in any part of the world where there are hostilities (whether war is declared or not) or enter a zone which is declared a war zone by any government or by the Insurers ;

- 7.1.13 ensure that where the Customer is not operating the Aircraft, that the operation of the Aircraft is governed by an **Operating Agreement** and where required by Lombard registered with the International Registry;
- 7.1.14 document any commercial use of the Aircraft on market standard terms;

7.1.15 perform and discharge all duties, liabilities or obligations imposed on the Customer under any Charter, Operating Agreement or other contract relating to the Aircraft;

7.1.16	if requested by Lombard, deposit with Lombard all deeds and documents of title, leases or Charters relating to the Security Assets other than those documents that by operation of applicable law must remain on the Aircraft;
7.1.17	not create or allow to subsist, or permit or permit any other person to create or grant, any Encumbrance over the Security Assets or any part thereof and shall procure the immediate discharge of any Encumbrance which arises or exists in respect of the Security Assets or any part thereof other than:
	(a) a Permitted Lien in respect of the Aircraft; or
	(b) any other Encumbrance disclosed to Lombard in writing prior to the date of the Aviation Loan Agreement and confirmed in the written confirmation provided under the Aviation Loan Terms;
7.1.18	not assign or purport to assign any Earnings or Insurances in relation to the Aircraft other than under the Finance Documents;
7.1.19	not amend or end, or grant a waiver under, any contract mortgaged, charged, assigned or held on trust under this Deed;
7.1.20	promptly, upon becoming aware of the same, notify Lombard of any Encumbrance over the Aircraft;
7.1.21	if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fireproof metal plate which has inscribed thereon (in legible characters) the following statement:
	"This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first priority mortgage in favour of Lombard ";
7.1.22	if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fireproof metal plate which has inscribed thereon (in legible characters) the following statement:
	"This Engine is owned by TAG Aviation (Stansted) Ltd and subject to a first priority mortgage in favour of Lombard ";
7.1.23	not sell, transfer or otherwise dispose of the Aircraft or any part thereof or interest therein other than with the prior written consent of Lombard (which consent Lombard shall have absolute discretion to withhold);
7.1.24	promptly give written notice to Lombard of the occurrence of any Event of Default, Default, Total Loss or any condition, event or circumstance which, with the lapse of time, might constitute a Total Loss;
7.1.25	comply with any other requirements which Lombard notifies to the Customer;
7.1.26	retain a copy of any certificate issued by the relevant tax authorities evidencing that VAT has been paid in respect of the Aircraft on board the Aircraft;
7.1.27	if VAT has not been paid in respect of the purchase of the Aircraft by the Customer, ensure that VAT does not become payable in respect of the Aircraft;
7.1.28	take (at its own cost) all steps required from time to time by applicable law or by customary and established practice or as Lombard may reasonably request to protect and perfect Lombard's interest in the Aircraft as mortgagee;
7.1.29	procure that the Aircraft
	(a) on and at all times following the date falling six weeks after drawdown under the Aviation Loan Agreement (or if earlier, the date on which the Aviation Authority first issues a valid certificate of airworthiness for the Aircraft), possesses a valid certificate of airworthiness issued by the Aviation Authority ;
· · · · · · · · · · · · · · · · · · ·	(b) possess a valid certificate of registration issued by the Aviation Authority;
	(c) possess (if issued by the Aviation Authority) a valid annual airworthiness review certificate;
	(d) possesses a valid annual release to service issued by the Maintenance Performer; and
	(e) possesses a valid Air Operator's Certificate on which the Aircraft is listed where the Aircraft is used for commercial air transport or aerial work.
7.1.30	deliver a copy of each renewed or reissued certificate of airworthiness, certificate of registration, airworthiness review certificate and a valid annual release to service to Lombard promptly upon such renewal or each annual reissue of each certificate (if applicable) by the Aviation Authority ;

• • • • •

- 7.1.31 not, remove or permit the removal of any Part or any equipment belonging to the Aircraft (other than in accordance with clause 8.2) without Lombard's prior written consent; procure that no steps are taken to deregister the Aircraft from the register of aircraft, or this Deed from the register of 7.1.32 mortgages, maintained by the Aviation Authority without the prior written consent of Lombard (which consent Lombard shall have absolute discretion to withhold); 7.1.33 ... procure that the Aircraft is at all times insured in accordance with the requirements of clause 10; and following an Event of Default which is continuing the Customer shall forthwith, upon the request of Lombard but at no 7.1.34 cost to Lombard (and without prejudice to the rights of Lombard pursuant to the Deregistration Power of Attorney), execute all such documents or agreements and take all such steps as are within its power and as are required in order to de-register the Aircraft from registration with, and export the Aircraft from, the State of Registration. 8 Operational and maintenance undertakings The Customer will at all times during the Security Period (or, if applicable, shall procure that any Operator of the 8.1 Aircraft will): keep the Aircraft in good and airworthy condition and maintain and repair the Aircraft, the Engines and all Parts in 8.1.1 accordance with (i) the Maintenance Programme, (ii) all mandatory rules and regulations of the Aviation Authority to the standard required by the Aviation Authority and (iii) any other regulations or requirements necessary in order to enable the Aircraft at any time to obtain a certificate of airworthiness issued by the Aviation Authority plus a valid annual release to service stamp issued by the Maintenance Performer; 8.1.2 maintain or cause to be maintained (in writing and in English) all logbooks, maintenance records, record books and other documents as are required by the Aviation Authority to the standard required by the Aviation Authority or the Maintenance Programme to be maintained with respect to the Aircraft; 8.1.3 replace or cause to be replaced promptly any Part of the Aircraft or any Engine in respect of which a Total Loss occurs with a replacement Part or Engine which shall be (i) owned by the Customer free and clear of all Encumbrances other than Permitted Liens (ii) subject to the Encumbrances intended to be created by this Deed and (iii) in such operating condition as, and of equivalent or better value to, the replaced Part or Engine was (or would have been if it had been maintained in accordance with the terms hereof), and where a Part or Engine is replaced under this clause, the Customer agrees to enter into a mortgage supplement in respect of such replacement Part or Engine; make or cause to be made all mandatory or recommended alterations and modifications and additions to the Aircraft 8.1.4 and the Engines as may from time to time be required by the Aviation Authority within the time period required by the Aviation Authority or any other relevant authority or the manufacturer of the Aircraft or any Engine; 8.1.5 comply with all mandatory Airworthiness Directives from time to time issued by the Aviation Authority or any other relevant authority and any modification recommended by the manufacturer of the Aircraft or any Engine in a mandatory service bulletin (unless such mandatory service bulletin is excluded by the Aviation Authority); ensure that all inspections (in connection with maintenance), services, repairs, overhauls, tests and modifications to be 8.1.6 made or carried out to or on the Aircraft are made or carried out by a maintenance performer approved by the Aviation Authority; 8.1.7 not suffer any arrest, detention, exercise of a possessory lien, execution, distress or other legal process to be levied or sued out against the Aircraft and in the event of an arrest, detention, exercise of a possessory lien, execution, distress or other legal process as aforesaid, the Customer shall notify Lombard immediately and shall take all steps necessary to ensure or procure that the Aircraft is promptly released therefrom; ensure that such crew and engineers as are employed in connection with the operation and maintenance of the Aircraft 8.1.8 have such qualifications and hold such licences as are required by the Aviation Authority and by all other applicable laws and regulations; operate the Aircraft or cause the Aircraft to be operated only in the ordinary course of its business and not operate or 8.1.9 permit the Aircraft to be operated for testing (other than for maintenance), training (other than of in-house pilots), racing, experimental flying or any other purpose for which it is not designated or reasonably suitable;
 - 8.1.10 at any time upon giving reasonable notice make available to Lombard for inspection the Technical Records and other log books of the Aircraft and all other repair, maintenance, modification, overhaul or other records kept by the Customer or any other person in respect of the Aircraft; and
 - 8.1.11 on request, promptly furnish to Lombard all such information regarding the operation, maintenance and condition of the Aircraft as Lombard may reasonably request.

Until the occurrence of an Event of Default, the Customer may:

8.2

- 8.2.1 remove any **Part** (other than an **Engine**) in the ordinary course of maintenance, service, repair, ove rhaul or testing, if it promptly replaces it with a replacement **Part** which satisfies the requirements of clause 8.1.3;
- 8.2.2 remove any **Engine** in the ordinary course of maintenance, service, repair, overhaul or testing provided that such **Engine** is re-installed on the **Aircraft**, or a replacement engine which satisfies the requirements of clause 8.1.3 is installed on the **Aircraft**, within thirty days (or such longer period as may be permitted by **Lombard**) of the removal from the **Aircraft** of such **Engine** and provided further that the removed **Engine** remains the property of the **Customer** unless and until a replacement engine satisfying the requirements of clause 8.1.3 is installed on the **Aircraft**;
- 8.2.3 make such alterations and modifications in and additions to the **Aircraft** as it thinks desirable in the proper conduct of its business or which may arise as a result of an **Airworthiness Directive** or a service bulletin provided that no such alteration, modification or addition (other than an Airworthiness Directive or mandatory service bulletin) breaches the requirements or recommendations of the **Aviation Authority**, the **Maintenance Programme**, the manufacturer or any other relevant authority or materially diminishes the value or utility of the **Aircraft** below that prevailing immediately prior to such modification or alteration or impairs the airworthiness of the **Aircraft** and further provided that no such alteration or modification may be made with or comprising **Parts** which are not in the absolute legal and beneficial ownership of the **Customer** free from all **Encumbrances** other than **Permitted Liens**;
- 8.2.4 remove any Part (other than an Engine) from the Aircraft without replacing it as required pursuant to the terms hereof if such Part (i) is in addition to and not in replacement of any Part previously installed on or attached to the Aircraft, (ii) is not required to be installed on or attached to the Aircraft by the terms hereof and (iii) can be removed from the Aircraft without impairing the airworthiness of the Aircraft or materially diminishing the value of the Aircraft below the value which the Aircraft would have had had such additional Part not been installed thereon or attached thereto;
- 8.2.5 whilst an **Engine** has been removed from the **Aircraft** in accordance with the provisions of clause 8.2.2, install a replacement engine notwithstanding that such replacement engine is not the property of the **Customer** provided that the removed **Engine** is re-installed, or such other engine as may satisfy the requirements of clause 8.1.3 is installed, on the **Aircraft** within the period specified in clause 8.2.2 and, further provided that the **Engine** substituted or replaced by such replacement engine shall at all times remain free from any **Encumbrance** other than **Permitted Liens**, shall remain the property of the **Customer**, and shall not cease to be subject to this **Deed** unless and until it is replaced by an engine which satisfies the requirements of clause 8.1.3; and
- 8.2.6 print or paint its name or other symbol on the Aircraft.
- 8.3 Save insofar as expressly required or permitted by the terms hereof and subject in any case as provided in clause 8.2, the **Customer** shall not, without the prior written consent of **Lombard** make any alteration in or modification or addition to the **Aircraft**, remove any part from the **Aircraft** or install any **Part** on or attach any **Part** to the **Aircraft** or permit any of the said events to occur.
- 8.4 The Customer shall promptly notify Lombard:
- 8.4.1 of any material alterations in or material modifications or additions to the Aircraft;
- 8.4.2 of any Encumbrance (excluding Permitted Liens) arising on the Aircraft and (in the case of all liens including Permitted Liens) exercised over the Aircraft or any part thereof;
- 8.4.3 (if Lombard so requests) the location of the Aircraft and of any Engine which is not, for the time being, installed on or attached to the Aircraft.
- 8.5 The existence of any agreement for maintenance of the **Aircraft** shall not in any way absolve, negate, reduce or mitigate the **Customer's** obligations to maintain the **Aircraft** in the manner required under this **Deed**.
- 8.6 The Customer shall:
- 8.6.1 ensure that the **Aircraft** shall not be used or operated when the **Insurances** are not in full force and effect or otherwise at a time or in a manner or in any geographical area which constitutes a breach of the terms of the **Insurances** or is otherwise inconsistent with the terms of the **Insurances**;
- 8.6.2 not cause or permit the Aircraft to be operated in contravention of any airworthiness certificate or the Aircraft operating manual relating thereto for the time being;
- 8.6.3 not make, do, permit, consent or agree to any act or omission to act (save as permitted or contemplated by this Deed) which would or might cause the Aircraft, or any Part thereof, to be exposed to penalty, forfeiture, impounding, destruction or abandonment under any applicable law or regulation;

8.6.4	not hold out or represent or permit the holding out or representation that any member of the crew of the Aircraft is in the employment of Lombard or that Lombard is carrying persons or goods in the Aircraft or are in any way connected or associated (whether for hire or gratuitously) with any operation of the Aircraft undertaken (or on behalf of) by the
	Customer;
8.6.5	not pledge the credit of Lombard for any maintenance, service or repairs to, or replacement or overhaul, of the Aircraft or any Engine or Part; and
8.6.6	pay and discharge, or cause to be paid and discharged, when due and payable, or make adequate provision by way of security or otherwise for all debts, damages, claims and liabilities which may have given or may reasonably be expected to give rise to any Encumbrance (other than a Permitted Lien) over or affecting the Aircraft.
9	Possession, subleasing and chartering
9.1	The Customer may not, without Lombard's prior written consent, charter the Aircraft.
9.1	The customer may not, without combard's phor written consent, charter the Aircrait.
9.2	The Customer may deliver possession of the Aircraft, Airframe, any Engine or any Part to the relevant Maintenance Performer or to any other suitably qualified maintenance contractor approved by Lombard for the purpose of performing maintenance or other work required or permitted by this Deed.
9.3	Save as permitted by clauses 9.1 and 9.2, the Customer shall not lease, or otherwise part with possession of, the Aircraft without the prior written consent of Lombard , which consent Lombard may grant or withhold in its absolute discretion.
9.4	If, pursuant to clause 9.1 or 9.3 Lombard consents to the Aircraft being leased or subleased, or otherwise operated by any other person, the Customer shall procure that:
9.4.1	the rights of that person under that lease, sublease or other operating agreement are expressed to be subject and subordinate to the rights of Lombard under this Deed;
9.4.2	that person executes and delivers to Lombard (prior to the commencement of that lease, sublease or other operating agreement) a Deregistration Power of Attorney, an Authority Letter and a consent and subordination agreement in terms specified by Lombard;
9.4.3	the lease, sublease or other operating agreement contains provisions which are at least as onerous as those set out in clauses 8 and 10;
9.4.4	the lease, sublease or other operating agreement contains such terms that shall not cause the Customer to be in breach of any Finance Document or cause an Event of Default;
9.4.5	the rights under that lease, sublease or other operating agreement (and any rights in relation to the insurances and warranties relating to the Aircraft, the Airframe or any Engine or Part) are assigned to Lombard (prior to the commencement of that lease, sublease or other operating agreement) in terms specified by Lombard.
10	Insurance undertakings
10.1	The Customer will at all times during the Security Period effect and maintain (or cause to be effected and maintained) at its own expense in relation to the Aircraft:
10.1.1	"Hull All Risks" insurance for loss of or damage to the Aircraft whilst flying and on the ground on an agreed value basis in an amount in the Required Insurance Currency which is no less than the Required Insurance Value;
10.1.2	"Spares All Risks" insurance (including "Hull, War and Allied Perils Risk" to the fullest extent available) on the Engines and Parts while not installed on the Airframe (and not considered part of the aircraft for insurance purposes) of the Aircraft for their full replacement cost (unless otherwise insured on an agreed value basis) in the Required Insurance Currency;
10.1.3	"Hull, War and Allied Perils" insurance, being such risks excluded from the "Hull All Risks" policy by application of War, Hijacking and Other Perils Exclusion Clause AVN48B (or subsequent version) to the fullest extent available from the leading aviation insurance markets, including confiscation and requisition by the State of Registration on an "agreed value basis" for an amount in the Required Insurance Currency which is no less than the Required Insurance Value .
10.2	The insurances required under clause 10.1 shall be provided on an agreed value basis (that is to say on the basis that in the event of a total loss the Insurers will pay the specified agreed value rather than the cost of acquiring a replacement aircraft) or as otherwise set out in this Deed and the policies shall include the form of AVN67B endorsement and, to the

10.2.1 include Lombard and its successors and assigns as additional insured and contract party for its rights and interests;

extent not in conflict with AVN67B:

10.2.2 include a loss payable clause that provides that all insurance proceeds in respect of a Total Loss shall be payable to Lombard;

- 10.2.3 include a reference to the Aviation Loan Agreement and this Deed;
- 10.2.4 include, in the event of separate insurances being arranged to cover the "Hull All Risks" insurance and the "Hull War Risk" and related insurance, a 50/50 claims funding arrangement in the event of any dispute as to whether a claim is covered by the "Hull All Risks" or "Hull War Risks" policy, wording in the terms of Lloyds clause AVS103 being sufficient for such purpose; and
- 10.2.5 be subject to a deductible (except in the event of Total Loss) of not more than the Permitted Deductible.

10.3 The **Customer** shall effect and maintain (or cause to be effected and maintained) at its own expense in relation to the **Aircraft**, a policy or policies of insurance covering aircraft third party, passenger, baggage, cargo, mail and airline general third party (including products, premises and hangerkeepers) legal liability for a combined single limit (bodily injury/property damage) of not less than the **Minimum Liability Cover** (and in any event no less than the amount carried in respect of the same or similar aircraft in the **Customer's** fleet in the same jurisdiction as the **Aircraft**). War and Allied Perils Risks shall also be covered in accordance with the then current market practice for aircraft of the same type as the **Aircraft**.

- 10.4 The policies evidencing the insurances required under clause 10.3 shall:
- 10.4.1 include Lombard and its officers, directors, employees, agents and its successors and assigns each as additional insured and as contract parties;
- 10.4.2 not conflict with AVN 67B;
- 10.4.3 provide that all the provisions thereof, except the limits of liability, shall operate to give each additional insured the same protection as if there were a separate policy covering each named insured save that this provision shall not operate to include any claim howsoever arising in respect of loss or damage to the **Aircraft** or relevant spare parts covered under the Hull or spares insurance, it being understood that notwithstanding the foregoing the total liability of **Insurers** in respect of any and all insured parties shall not exceed the limits of liability stated in the policy; and
- 10.4.4 be primary and without right of contribution from other insurance which may be available to **Lombard** or any of its officers, directors, employees, agents and its successors and assigns.
- 10.5 The policies evidencing any Insurances with respect to the Aircraft required under clauses 10.1 and 10.3 shall:
- 10.5.1 provide that the cover afforded to the additional insured shall not be invalidated by any act or omission (including misrepresentation and non-disclosure) of any other person or party which results in a breach of any term, condition or warranty of the policy provided that the additional insured so protected has not caused, contributed to or knowingly condoned the said act or omission;
- 10.5.2 provide for worldwide coverage (subject only to such exceptions as are consistent with generally accepted market practice from time to time or such other exceptions as **Lombard** may reasonably agree in writing);
- 10.5.3 provide that the **Insurers** shall waive any right of subrogation to any right against **Lombard**, any of its officers, directors, employees, agents and its successors, transferees and assigns;
- 10.5.4 provide that Lombard shall not be liable for any premiums in respect thereof and that the Insurers shall not exercise any right of set-off or counterclaim in respect of unpaid premiums or otherwise against the interest of Lombard otherwise than in respect of unpaid premiums with respect to the Insurances of the Aircraft; and
- 10.5.5 provide that the required insurances may only be cancelled or materially altered in a manner adverse to the additional insureds and contract parties by the giving of not less than thirty days' notice in writing (but seven days or such lesser period as may be customarily available in respect of War and Allied Perils) to **Lombard**. Notice shall be deemed to commence from the date such notice is given by the **Insurers**. It is understood that such notice will not, however, be given at normal expiry date of the policy or any endorsement.
- 10.6 All **Insurances** required under clauses 10.1 or 10.3 shall be maintained with such insurance underwriters and/or insurance companies as **Lombard** may reasonably approve.
- 10.7 The Customer shall, in relation to the Aircraft, at its own cost and expense:
- 10.7.1 furnish or procure that there is furnished to Lombard, forthwith upon the date not later than seven days after the renewal date of each **Insurance**, a certificate or certificates, as hereinafter described, signed by the relevant **Insurers** (or their authorised representatives) or brokers, reasonably acceptable to Lombard, which have been appointed by the **Customer** (the "Brokers") providing evidence of insurance coverage pursuant to this clause 10. Each such certificate

shall (i) describe the perils covered by each policy of insurance issued by it then in force, (ii) specify the amounts of insurance coverage provided against each such peril and any deductible or uninsured amount and (iii) certify that each such policy is then in full force and effect;

- 10.7.2 furnish or procure that there is furnished to Lombard, on written request, evidence of (a) payment of each sum then due and payable under or in connection with any required Insurance and (b) before or on the expiry of any policy or policies in respect of such Insurances, renewal of the relevant policy or policies or that renewal will be effected on expiry; and
- 10.7.3 procure that the relevant **Insurers** or **Brokers** provide **Lombard** with such other information as to any **Insurances** as **Lombard** may reasonably request.
- 10.8 The **Customer** shall, in relation to the **Aircraft** and at its own cost and expense, cause the relevant **Brokers** to issue a written undertaking in favour of **Lombard** in such form as **Lombard** may reasonably approve having regard to market practice at such time which shall, subject as aforesaid, inter alia, (i) require such **Brokers** to notify **Lombard** forthwith upon such **Brokers** becoming aware of any cancellation or non-renewal of or modification to a required **Insurance** or of their ceasing to act as insurance brokers to the **Customer** or of any other fact or circumstance of which they may at any time learn which might invalidate any required insurance or render it wholly or partially unenforceable and (ii) require the **Brokers** to provide **Lombard** with a certificate of insurance at the same time as the **Broker** supplies the same to the **Customer**.
- 10.9 The Customer shall, in relation to the Aircraft, forthwith notify Lombard of any event (including but not limited to a Total Loss) which will or may give rise to a claim under any required Insurance.
- 10.10 The Customer shall not in relation to any required Insurance taken out in respect of the Aircraft:
- 10.10.1 make or permit the making of any modification to any required Insurance without the prior written consent of Lombard; or
- 10.10.2 do or omit to do or permit to be done or left undone anything whereby any required **Insurance** would or might reasonably be expected to be rendered in whole or in part invalid or unenforceable and, without prejudice to the foregoing, shall not use or keep or permit the **Aircraft** or any part thereof to be used or kept for any purpose, in any manner or in any place not covered by the required **Insurances**; or
- 10.10.3 cause or knowingly permit the Aircraft to be employed or used in any place or in any manner or for any purpose inconsistent with the terms of or outside the cover provided by any required Insurances or if the Insurances have lapsed.
- 10.11 The Customer shall, in relation to any required Insurance taken out in relation to the Aircraft, bear any part of any loss or liability which is to be borne by the insured under any required Insurance (notwithstanding that such Insurance names Lombard or other person as additional named insured).
- 10.12 The **Customer** shall, in relation to any required **Insurance** taken out in relation to the **Aircraft**, reimburse **Lombard** on demand the amount of any premiums or premium instalments which **Lombard** may pay together with interest thereon calculated for the period from the date of payment by **Lombard** until the date of payment by the **Customer** (after as well as before judgment).
- 10.13 If the **Customer** shall fail to maintain or cause to be maintained **Insurances** which in the opinion of **Lombard** meet the requirements hereof, **Lombard** may at its option, after notice in writing to the **Customer**, obtain such **Insurances** and in such event the **Customer** shall, upon demand, reimburse **Lombard** for the cost thereof together with interest thereon for the period from the date of payment by **Lombard** to the date of payment by the **Customer** (after as well as before judgment).
- 10.14 Without in any way limiting the **Customer's** obligations under the other provisions of this clause 10, the **Customer** undertakes that if, at any time, clause "AVN2000 Date Recognition Exclusion Clause" or any equivalent clause is endorsed on the policies of insurance required hereunder, the **Customer** shall, at its cost and expense, do all things and take any and all such action available to it to ensure that the interests of **Lombard** under such policies of insurance are fully protected (including, without limitation, ensuring that clauses "AVN2001 Date Recognition Limited Coverage Clause" and "AVN2002 Date Recognition Limited Coverage Clause" or any equivalent clauses are endorsed on the policies of insurance are fully protected hereunder as soon as possible thereafter). If at any time **Lombard** shall be of the opinion (acting reasonably) that the interests of **Lombard** in the insurances are not adequately protected by reason of the operation of clause "AVN2000 Date Recognition Exclusion Clause" or any equivalent clause, **Lombard** may (but shall not be obliged to) take out such policy of insurance to insure **Lombard** against such risks as **Lombard** may think fit (acting reasonably) and the **Customer** shall indemnify and does hereby indemnify **Lombard** against any and all costs incurred in connection therewith.

11 Requisition

11.1 The Customer will remain fully liable to Lombard for all its obligations under this Deed should there be a Requisition of the Aircraft (other than those with which it is physically unable to comply owing to the requisition) unless and until

Lombard is satisfied that the requisition authority has assumed the Customer's obligations to Lombard, and Lombard has so confirmed to the Customer.

- 11.2 The Customer agrees that the Requisition Compensation will be paid to Lombard by way of prepayment of the Customer's Obligations if Lombard requests.
- 11.3 When any Requisition ends the Customer will immediately and fully comply with all of its obligations under this Deed.

12 Indemnity to Lombard

- 12.1 The Customer will indemnify Lombard against any liability or loss (including legal costs) arising from any claims as a result of (a) the Customer or Lombard entering into this Deed; (b) the Customer failing to perform any of its obligations under this Deed; (c) Lombard investigating any event that it believes to be a default under this Deed; or (d) Lombard relying on any notice, request or instruction delivered under this Deed. The Customer will pay Lombard within three Business Days of demand by Lombard.
- 12.2 The Customer shall indemnify Lombard against any Taxation or similar charge imposed by any governmental or taxing authority which arises in relation to the Aircraft, this Deed or any payments due under this Deed (including for the avoidance of doubt, any amount of value added tax determined by Lombard to be irrecoverable but excluding Taxes on Lombard's overall net profits). The Customer will pay Lombard within three Business Days of demand by Lombard.
- 12.3 Where the **Customer** indemnifies or reimburses **Lombard** for any liability, loss or **Tax** the amount payable by way of indemnity or reimbursement shall take account of the likely **Tax** treatment in the hands of **Lombard** (as conclusively determined by **Lombard**) of (i) the indemnity payment or reimbursement and (ii) the **Tax** or loss suffered in respect of which that amount is payable. In the event that the assumed **Tax** treatment of the indemnity payment or any reimbursement proves to be incorrect the **Customer** agrees to pay, on demand, any additional sum required to ensure that **Lombard** occupies the same after-**Tax** position as it would occupy if no such adjustment were necessary.
- 12.4 These indemnities will continue in force notwithstanding any Event of Default or termination of this Deed.

13 When security becomes enforceable

Without prejudice to Lombard's other rights under this Deed or by virtue of applicable law, upon the occurrence of an **Event of Default**, and at any time thereafter, Lombard may, irrespective of whether a notice has been served on the **Customer**, declare the security constituted by the Security Documents to have become immediately enforceable and thereafter Lombard shall be entitled, without further notice or further demand, immediately to exercise all the powers and remedies possessed by it according to applicable law as mortgagee of the Aircraft and the other Security Assets and in particular but, without limitation, the powers set out in this Deed.

14 Possession and exercise of powers

- 14.1 Lombard does not have an immediate right to possession of the Aircraft or its income (and will not be considered to be taking possession if it enters to inspect or repair the Aircraft). The Customer will continue in possession until Lombard takes possession.
- 14.2 Any purchaser or third party dealing with **Lombard** or a receiver may assume that **Lombard's** powers have arisen and are exercisable without proof that demand has been made.
- 14.3 The receipt of Lombard or its delegate or any receiver shall be conclusive discharge to a purchaser in any sale or disposal of any of the Security Assets.

15 Appointment of receiver

Following the occurrence of an **Event of Default, Lombard** may appoint or remove a receiver or receivers of the **Aircraft**. Any restriction imposed by law on the right of a mortgagee to appoint a receiver (including under section 109(1) of the Law of Property Act 1925) that can be excluded does not apply to this **Deed**. If **Lombard** appoints a receiver, **Lombard** may agree and pay the receiver's fees and expenses (without being limited to the maximum rate specified in section 109(6) of the Law of Property Act 1925). The receiver will be the **Customer's** agent and the **Customer** (and not **Lombard**) will be responsible for the acts, defaults and remuneration of the receiver.

16 Powers of Lombard and receivers

- 16.1 When the security created by this **Deed** becomes enforceable in accordance with clause 13, **Lombard** or any receiver may:
- 16.1.1 enter, take possession of, operate and/or generally manage the Aircraft and enter upon any land or premises where the Aircraft may be located without being liable to the Customer (or any lessee of the Aircraft) for or by reason of such entry;

- 16.1.2 apply to any authority in the country in which the Aircraft or any part thereof is located for an enforcement and attachment order in respect of the Aircraft;
- 16.1.3 apply to any court of competent authority for an order so as to vest all the Customer's right, title and interest in the Aircraft in Lombard absolutely;

16.1.4 sell, call in, collect and convert into money the **Aircraft** to the extent that it is entitled to do so with all such powers in that respect as are conferred by applicable law (the power of sale shall arise, and the **Customer's Obligations** shall be deemed due and payable for that purpose, on the execution of this **Deed**) and by way of extension thereof such sale, calling in, collection and conversion may be made by public or private contract at any place in the world with or without notice to the **Customer**, without being liable to account for any loss of or deficiency in the consideration therefore and so that **Lombard** shall have power in its absolute discretion to postpone any such sale or to proceed to a forced sale without being liable for any loss resulting there from (except that resulting from **Lombard's** wilful misconduct or gross negligence), and for the purposes of this clause 16.1.4 **Lombard** may (in its sole discretion) by written notice appoint the **Customer** its non-exclusive agent to sell the **Aircraft** on terms satisfactory to **Lombard**;

- 16.1.5 maintain and/or repair the Aircraft;
- 16.1.6 settle, arrange, compromise or submit to arbitration any accounts, claims, questions or disputes whatsoever which may arise in connection with or in any way relating to the **Aircraft** or the **Customer's Obligations** and execute releases or other discharges in relation thereto;
- 16.1.7 put the Aircraft in storage for such term and generally in such manner and upon such conditions and stipulations as Lombard shall think fit;
- 16.1.8 execute and do all such acts, deeds and things as **Lombard** may consider necessary or proper for or in relation to any of the purposes aforesaid;
- 16.1.9 hire, lease, charter or accept surrenders of any hire, lease or charter, charge or deal with the Aircraft without restriction
- 16.1.10 take, continue or defend any proceedings and enter into any arrangement or compromise in relation to the Security Assets;
- 16.1.11 insure the Aircraft, arrange indemnity and other similar insurance, and obtain bonds and give counter-indemnities and other security in connection with this Deed;
- 16.1.12 employ advisers, consultants, managers, agents and others;
- 16.1.13 purchase or acquire materials, tools, equipment, furnishing, goods or supplies;
- 16.1.14 collect, recover and give good discharge for any moneys or claims forming part of, or relating to, any Security Assets and to permit any brokers through whom collection or recovery is effected to charge the usual brokerage for such collection or recover; and
- 16.1.15 do any acts which **Lombard** or a receiver considers to be incidental to the exercise of its powers or generally beneficial to **Lombard**.
- 16.2 A receiver may borrow and secure the repayment of any money, in priority to the Customer's Obligations.
- 16.3 In addition to the powers set out in this clause 16, a receiver shall have all the powers conferred on a receiver by the Law of Property Act 1925 and the Insolvency Act 1986 and any other applicable statute. Sections 93 and 103 of the Law of Property Act 1925 shall not apply to this **Deed**.
- 16.4 Joint receivers may exercise their powers jointly or separately.
- 16.5 A receiver will first apply any money received from the **Aircraft** towards the repayment of all money that the receiver has borrowed and secondly in payment of the receiver's fees and expenses. The receiver will then apply any remaining money received as required by law.
- 16.6 **Lombard** may exercise any of its powers whether or not a receiver has been appointed.
- 16.7 Each of **Lombard** and any receiver shall have full power to delegate (either generally or specifically) the powers, authorities and discretions conferred on it by this **Deed** on such terms and conditions as it sees fit, which delegation shall not preclude either the subsequent exercise of such power, authority or discretion by **Lombard** or the receiver itself or any subsequent delegation or revocation of it.

16.8	Any credit balance with Lombard will not be repayable, or capable of being transferred, charged or dealt with by the Customer until the Customer's Obligations have been paid in full.
16.9	None of Lombard , its delegate(s), nominee(s) or any receiver shall be liable for any loss by reason of (a) taking any action permitted by this Deed or (b) any neglect or default in connection with the Security Assets or (c) taking possession of or realising all or any part of the Security Assets , except in the case of gross negligence or wilful default upon its part.
16.10	Lombard will not be required in any manner to perform or fulfil any obligation of the Customer, make any payment, make any enquiry as to the nature or sufficiency of any payment received by it or present or file any claim or take any action to collect or enforce the payment of any amount.
16.11	Lombard shall not be liable either to the Customer or to any other person by reason of the appointment of a receiver or for any other reason.
16.12	Neither Lombard nor the receiver will be in any way liable or responsible to the Customer for any loss or liability arising from any act, default, omission or misconduct on the part of any delegate or sub-delegate.
17	Application of payments
17.1	Lombard may apply any payments received for the Customer to reduce any of the Customer's Obligations, as Lombard decides.
17.2	Until all amounts which may be or become payable by the Customer to Lombard have been irrevocably paid in full, Lombard (or any trustee or agent on its behalf) may without affecting the liability of the Customer under this Deed :
17.2.1	refrain from applying or enforcing any other monies, security or rights held or received by Lombard (or any trustee or agent on its behalf) against those accounts or apply and enforce them in such manner and order as it sees fit (whether against those amounts or otherwise); and
17.2.2	hold in an interest-bearing suspense account any moneys received from the Customer or on account of the Customer's liability under this Deed .
17.3	If Lombard reasonably considers that any amount paid or credited to it is capable of being avoided or reduced by virtue of any bankruptcy, insolvency, liquidation or similar laws, the liability of the Customer under this Deed and the security created under this Deed shall continue and such amount shall not be considered to have been irrevocably paid.
17.4	If Lombard receives notice of any Encumbrance affecting the Aircraft other than a Permitted Lien, Lombard may suspend the operation of the Customer's account(s) and open a new account or accounts. Regardless of whether Lombard suspends the account(s), any payments received by Lombard after the date of that notice will be applied first to repay any of the Customer's Obligations arising after that date.
18	Preservation of other security and rights and further assurance
18.1	This Deed is in addition to any other security for the Customer's Obligations held by Lombard now or in the future. Lombard may consolidate this Deed with any other security so that they have to be redeemed together, but it will not merge with or prejudice any other security or Lombard's other rights.
18.2	The security created under this Deed will be a continuing security for the ultimate balance of the Customer's Obligations notwithstanding any intermediate payment or settlement of all or any part of the Customer's Obligations .
18.3	The obligations of the Customer under this Deed will not be affected by any act, omission or thing which, but for this provision, would reduce, release or prejudice any of its obligations under this Deed (whether or not known to it or Lombard). This includes:
18.3.1	any time or waiver granted to, or composition with, any person;
18.3.2	any release of any person under the terms of any composition or arrangement;
18.3.3	the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce any rights against, or security over assets of, any person;
18.3.4	any non-presentation, non-observance of any formality or other requirements in respect of any instrument or any failure to realise the full value of any security;
18.3.5	any incapacity, lack of power, authority or legal personality of or dissolution or change in the members or status of any person;
18.3.6	any amendment of any document or security:

iy uocumer н ог security; y

• • • •

-

- 18.3.7 any unenforceability, illegality, invalidity or non-provability of any obligation of any person under any document or security or the failure by any person to enter into or be bound by any document or security; and
- 18.3.8 any insolvency or similar proceedings.
- 18.4 Without prejudice to the generality of clause 18.2 above, the **Customer** expressly confirms that it intends that the security under this **Deed** shall extend from time to time to any variation, increase, extension or addition (however fundamental) of or to any document creating any of the **Customer's Obligations** and/or any facility or amount made available under such document.
- 18.5 Unless all amounts which may be or become payable by the **Customer** to **Lombard** have been irrevocably paid in full or as **Lombard** otherwise directs, the **Customer** will not, after a claim has been made or by virtue of any payment or performance by it under this **Deed**:
- 18.5.1 be subrogated to any rights, security or moneys held, received or receivable by Lombard (or any trustee or agent on its behalf); nor
- 18.5.2 be entitled to any right of contribution or indemnity in respect of any payment made or moneys received on account of the **Customer's** liability under this clause.
- 18.6 The **Customer** waives any right it may have of first requiring **Lombard** (or any trustee or agent on **Lombard's** behalf) to proceed against or enforce any other right or security or claim payment from any person or file any proof or claim in any insolvency, administration, winding-up or liquidation proceedings relative to any other person before claiming from the **Customer** under this **Deed**.
- 18.7 If the Customer's Jurisdiction of Incorporation, or the State of Registration has, or at any time brings into force, any legislative or other provisions giving effect to the Cape Town Convention and Aviation Protocol, the Customer shall:
- 18.7.1 promptly register together with Lombard any interest with the International Registry with respect to the Aircraft and/or any Engine and constituted by any Finance Document;
- 18.7.2 promptly register together with Lombard any interest with the International Registry with respect to any replacement Engine provided in accordance with clause 8.1.3 at the time of such replacement;
- 18.7.3 promptly do all acts and things and enter into any agreements (subordination or otherwise) reasonably required by **Lombard** to protect and/or enhance and/or improve the title, rights, priority of any international interests and any other interests vested in **Lombard** with respect to the **Aircraft** and/or any **Engine** and constituted by any **Finance Document** and consent to the registration of any of the foregoing;
- 18.7.4 agree to any amendment, restatement, revision or any variation to any Finance Document to the extent necessary if required by Lombard to permit the interests created by the relevant Finance Document to constitute international interests under the Cape Town Convention and Aircraft Protocol;
- 18.7.5 promptly execute and deliver an irrevocable de-registration and export request authorisation in respect of the Aircraft in the form set out Schedule 6 (an "IDERA");
- 18.7.6 use it best endeavours to promptly procure countersignature of the IDERA by the Aviation Authority and return an original IDERA signed by the Customer and the Aviation Authority to Lombard; and
- 18.7.7 not register or consent to the registration of, any conflicting interests (whether or not taking priority over Lombard's international interests) at the International Registry without Lombard's prior written consent.
- 18.8 The **Customer** shall not change the State of Registration without the consent of Lombard (which Lombard shall have absolute discretion to withhold). At the request of Lombard upon a change of the State of Registration of the Aircraft taking place, the Customer shall execute in favour of Lombard a deed amending or supplementing this Deed and/or a mortgage and/or any other document required by Lombard for the purpose of preserving the security intended to be created hereby in such form as Lombard may reasonably request.

19 Non-compliance by the Customer

If the **Customer** for any reason fails to observe or punctually perform any of its obligations under this **Deed**, **Lombard** shall be entitled, on behalf of or in the name of the **Customer** or otherwise and at the cost of the **Customer**, to perform the obligation and to take any steps which **Lombard** may consider necessary with a view to remedying or mitigating the consequences of the failure.

20 Power of attorney

. .

By way of security, to give effect to this **Deed** and to perfect, protect and enhance the security created by this **Deed** and secure the exercise of any of their powers, the **Customer** irrevocably appoints **Lombard**, and separately any receiver appointed pursuant to the provisions of this **Deed**, to be the **Customer's** attorney (with full power of substitution and delegation), in the **Customer's** name to sign or execute any documents, deeds and other instruments, or take, continue or defend any proceedings in accordance with this **Deed** and the **Aviation Loan Agreement**. The **Customer** ratifies and confirms whatever any attorney does or purports to do pursuant to its appointment under this clause 20.

21 Lombard's rights

If Lombard waives or decides not to enforce any of its rights under this Deed, Lombard may enforce any such right at a later date.

22 Set-off

Lombard may at any time, and without notice to the Customer, combine or consolidate all or any of the Customer's then existing accounts with any liabilities to Lombard and apply any sum due to the Customer in satisfaction of any sum due and unpaid by the Customer under any Finance Document or any other agreement, or guarantee with or other obligation owed to Lombard.

23 Currency

- 23.1 **Lombard** may convert any payment it receives in any other currency in respect of this **Deed** to the **Currency** at its chosen spot rate of exchange for the purchase of the relevant currency in the London foreign exchange market at the relevant time.
- 23.2 Lombard may convert the Value stated in any valuation supplied to it in connection with this Deed from one currency to the Currency, at its chosen spot rate of exchange for the purchase of the relevant currency in the London foreign exchange market at the relevant time.

24 Severability

If at any time any provision in connection with this **Deed** becomes illegal, invalid or unenforceable the remaining provisions will not in any way be affected or impaired thereby.

25 Variations

No variation of these terms shall be valid unless it is in writing and signed by or on behalf of each of the Customer and Lombard.

26 Notices

- 26.1 All consents, notices and demands relating to this Deed must be in writing (which shall include by electronic means).
- 26.2 **Lombard** may deliver a notice or demand to the **Customer** at the contact details last known to **Lombard**, the **Customer's** registered office, or the **Address for Service**, if one is specified.
- 26.3 A notice or demand by **Lombard** will be deemed given at the time of personal delivery; on the next **Business Day** after posting; or, if by fax or other electronic means, at the time of sending, if sent before 6.00 pm (London time) on a **Business Day**, or otherwise on the next **Business Day**.
- 26.4 The Customer must deliver a notice to Lombard addressed to the address or email address shown above. Notice by the Customer will be effective when received by Lombard, provided that any email received on any day other than a Business Day, or at any time after 6.00pm (London time) on a Business Day, shall not be effective until 9.30am (London time) on the next Business Day.

27 Transfers

27.1 Lombard may allow any person to take over any of its rights or duties under this Deed and may provide information about the Customer or any Finance Document to which the Customer is party to the transferee in connection with any proposed transfer and any person to whom such information is required to be disclosed by any applicable law.

27.2 The Customer is not entitled to assign or otherwise transfer its rights or obligations under this Deed.

28 Counterparts

This Deed may be executed in any number of counterparts and all such counterparts taken together will be deemed to constitute one and the same instrument.

9	Third party rights	
	Unless expressly provided to the contr	ary in this Deed, a person who is not a party to this Deed has no right under the
	Contracts (Rights of Third Parties) Act	1999 to enforce or to enjoy the benefit of any term of this Dead.
0	· Entire careement	
V	Entire agreement	
	This Deed (and any agreed variations herein, sets out the entire agreement relation to the subject matter of the De	made in accordance with the terms of this Deed) and any documents referred to between the parties. It supersedes any previous agreement between them in ed.
1	Customer Information	
	Appendix 1 details who Lombard may	share Customer information with and for what purpose(s).
2	Law	
- 		al de la servicie de La servicie de la serv La servicie de la serv
	inis Deed and any non-contractual ob	ligations arising out of or in connection with it are governed by English law.
2.1	 irrevocably agrees that a judgment or rule and binding on the Customer and may 	mer irrevocably submits to the non-exclusive jurisdiction of the English courts and uling in any proceedings connected with the Deed in those courts will be conclusive y be enforced against the Customer in the courts of any other jurisdiction. To the ake concurrent proceedings in any number of jurisdictions.
2.2	The Address for Service, if specified, serving proceedings on the Customer.	, or any other address provided for this purpose, will be an effective address for .
Exec	uted and Delivered as a deed by TAG A	viation (Stansted) Ltd
Exec	uted and Delivered as a deed by TAG A	viation (Stansted) Ltd
Exec	uted and Delivered as a deed by TAG A	Dire
Exec		
Di	TREVOR Gunn name	
Di	TREVOR GUNN	Dire Dire
Di	TREVOR GUNN	Dire Dire
Print	TREVOR GUNN	Dire Dire Print name
Print	TREVOR Gunname	Dire Dire LOUISC GUNN Print name
Print	TREVOR GUNN	Dire Dire LOUISC GUNN Print name
Print If only Signe	TREVOR Gunname name y one signature is required, it must be that ad by the Director in the presence of:	Dire Dire Print name At of a Director, and a witness is required
Din Print If only Signe	TREVOR Gummaname y one signature is required, it must be that ad by the Director in the presence of: ass' signature	Dire Dire ADUISCANN Print name
Din Print If only Signe Witne	TREVOR Guve name y one signature is required, it must be that ad by the Director in the presence of: ess' signature ess' name in full	Dire Dire Print name At of a Director, and a witness is required
Din Print If only Signe Witne	TREVOR Gummaname y one signature is required, it must be that ad by the Director in the presence of: ass' signature	Dire Dire Print name At of a Director, and a witness is required
Din Print If only Signe Witne	TREVOR Guve name y one signature is required, it must be that ad by the Director in the presence of: ess' signature ess' name in full	Dire Dire Print name At of a Director, and a witness is required
Dia Print If only Signe Witne Addre	TREVOR Guve name y one signature is required, it must be that ad by the Director in the presence of: ess' signature ess' name in full	Dire Dire Print name
Dia Print If only Signe Witne Addre	TREVOR Guess' signature is required, it must be that ad by the Director in the presence of:	Dire Dire Print name
Dia Print If only Signe Witne Addre	TREVOR Guess' signature is required, it must be that and by the Director in the presence of:	Dire Dire Print name
Dia Print If only Signe Witne Addre	TREVOR Guess' signature is required, it must be that ad by the Director in the presence of:	Dire Dire Print name
Print Print Signe Witne Addre Signe	TREVOR Guess name y one signature is required, it must be that ad by the Director in the presence of: ess' signature	Dire Dire Print name
Print Print Signe Witne Addre Signe	TREVOR Guess' signature is required, it must be that and by the Director in the presence of:	Dire Dire Print name
Print Print Signe Witne Addre Signe	TREVOR Guess name y one signature is required, it must be that ad by the Director in the presence of: ess' signature	Dire Dire Print name
Print Print Signe Witne Addre Signe	TREVOR Guess name y one signature is required, it must be that ad by the Director in the presence of: ess' signature	Dire Dire Print name
Print Print Signe Witne Addre Signe	TREVOR Guess name y one signature is required, it must be that ad by the Director in the presence of: ess' signature	Dire Dire NOUISCANN Print name

Original	Copy -	· Page	1	of	2
----------	--------	--------	---	----	---

Aircraft Mortgage

THIS IS AN IMPORTANT DEED AND SHOULD ONLY BE USED WHERE THE AIRCRAFT IS ON THE GROUND IN ENGLAND OR IN ENGLISH AIRSPACE AT THE TIME THIS DEED IS CREATED UNLESS FOREIGN LEGAL ADVICE CONCERNING LEX SITUS IS OBTAINED. YOU SHOULD TAKE LEGAL ADVICE BEFORE SIGNING.

Data				۵۰ ۱۹۰ ۱۹۰۰
Date	b June	2021	(to be completed by Lombard)	
Parties Lombard North Central F	PLC, a company re	gistered in England and Wales	(registered number: 00337004)	
whose address for notices 8778888 Text Relay: 18001 0345 8		ithernam, S63 3BR. Email: ge	neralenquiries@lombard.co.uk. Cu	stomer service: 0345
Customer's name	TAG Aviation (Stansted) Ltd		
Customer's address	The Diamond H	langar, Long Border Road, S	itansted Airport, Essex, England	. CM24 1RE
		NYYYY MARAA MAMINY MUNYY IN MALALAM Y MALALAMIN MUMALALAMIN' MUMALALAMIN' MUMALALAMIN' MUMA	Имелличических лакималичичие личносок сокаличиство с с с с с с с с с с с с с с с с с с с	
Customer's Jurisdiction	of Incorporation	England and Wales	алаананаа алаа алаа алаа алаа алаа алаа	
Company / LLP Registra	tion number	06137634		
Address for Service (must be in the United Kingdom and include name)	The Diamond H	langar, Long Border Road, S	tansted Airport, Essex, England	I, CM24 1RE
Telephone Number:	01279 815544	9999-9999	Fax Number: N/A	
Contact:	Mr Trevor Gunn		elenet foren. Meneterine Meneterinen er en en en er en er	je. Nastrona (nastrona) Nastrona (nastrona)
Is the Customer a SPV?	No			
Operator				· 700-710-711-711-711-711-711-711-711-711-
Name:	N/A		Company / LLP Registration number:	N/A
Operator address:	N/A			······································
			24. A second s Second second sec Second second s Second second s Second second seco	
	an a			

LAFG/MAA/---/1018 (T)

Lombard

Original Co	οpy »	Page	2	of	2
-------------	-------	------	---	----	---

Aircraft Details	
Registration Mark	G-NEWG
Aircraft Manufacturer	The Boeing Company
Aircraft Type	Boeing 737-700
Year of Manufacture	2002
Manufacturer's Serial Number	29885

	Quantity	Manufacturer / Type	Serial Numbers
Engines		CFM / CFM56 -7B22	888920 & 888940
Avionics			
Propeller / Rotor blades			
APU			

in the first state of the second state of the	
State of Registration	United Kingdom
Aviation Authority	UK Civil Aviation Authority
Maintenance Performer	N/A
Habitual Base	Lasham Airfield
Required Insurance Value	The greater of 110% of the balance outstanding or the current market value of the aircraft

	ancian		
Minimum Liability Cover	£322,000,000.00	Permitted Deductible	
Major Damage Sum	£10,000.00		an a

(a) A set of the se	na na shipha a shipha a shi	 A set of the set of	1	1.1
Each Engine is 550 or more rate	d takeoff horsepower	or its equivalent	Yes	÷.,

Each Propeller is capable of absorbing 750 or more related takeoff shaft horsepower | N/A

The Airframe is described on the International Registry Manufacturer's List as BOEING model 737-700 with serial number 29885

The Engine(s) is/are described on the International Registry Manufacturer's List as CFM model CFM56-7 with serial number 888920 & 888940

Aircraft Mortgage (CAA)

LAFG/MAA/---/1018 (T)

·		
1 De	finitions and interpretation	
1.1 In	this Deed the following words will have th	e meanings given alongside them:
1.1.1	Aircraft	many the stratest more particularly densitied in the Alexande details
3. 1. 1	Aircran	means the aircraft more particularly described in the Aircraft details together with the Engines and Propellers (whether or not any of the Engines and Propellers may from time to time be installed on the Airframe), the Technical Records and all Parts now or hereafter installed in or on the Airframe and all substitutions, renewals and replacements from
		time to time made in or to or installed in or on the Airframe in accordance with this Deed including any Parts which are for the time being detached from the Airframe but which remain the property of the Customer ;
1.1.2	Airframe	means the Aircraft excluding the Engines and the Technical Records;
1.1.3	Airworthiness Directive	means an airworthiness directive issued by the Aviation Authority or by EASA or the national civil aviation authority of the Aircraft's country of manufacture;
1.1.4	Authorisation	has the meaning given to it in the Aviation Loan Terms;
1.1.5	Authority Letter	the letters sent or to be sent by the Customer to any party (including Eurocontrol) in respect of any charges (including navigation charges) in relation to the Aircraft in the form set out in schedule 5;
1.1.6	Aviation Authority	each person who shall from time to time be vested with the control and supervision of, or have jurisdiction over, the registration, airworthiness and operation of aircraft or other matters relating to civil aviation in the State of Registration;
1.1.7	Aviation Loan Agreement	the aviation loan agreement entered into between the Customer and Lombard in relation to the financing of the Aircraft inclusive of the Aviation Loan Terms;
1.1.8	Aviation Loan Terms	the aviation loan terms which form part of the Aviation Loan Agreement;
1.1.9	Business Day	has the meaning given to it in the Aviation Loan Terms;
1.1.10	Cape Town Convention and Aircraft Protocol	means the Convention on International Interests in Mobile Equipment and the Aircraft Protocol on International Interests in Mobile Equipment or Matters specific to Aircraft Equipment each signed at Cape Town on the 16th November, 2001;
1.1.11	Charging Period	has the meaning given to it in the Aviation Loan Terms;
1.1.12	Charter	any charter relating to the Aircraft or any other contract for use o employment of the Aircraft pursuant to clause 9;
1.1.13	Charter Guarantee	any guarantee, bond, letter of credit or any other form of security supporting a Charter;
1.1.14	Compulsory Acquisition	means, in respect of the Aircraft, the Airframe and/or an Engine requisition for title or other compulsory acquisition of title (but excluding requisition for use or hire) of such Aircraft, Airframe or Engine, as the case may be;
1.1.15	Currency	has the meaning given to it in the Aviation Loan Agreement;
1.1.16	Customer's Obligations	has the meaning given to it at clause 2;
1.1.17	Default	has the meaning given to it in the Aviation Loan Terms;
1.1.18	Deregistration Power of Attorney	a power of attorney to deregister the Aircraft in favour of Lombard in the form set out in Schedule 1;

1.1.19	EASA	the European Aviation Safety Authority and any successor organisation or authority;
1.1.20	Earnings	all amounts of any kind which may at any time be earned, due or payable arising from the use, ownership, possession or operation of the Aircraft;
1.1.21	Encumbrance	has the meaning given to it in the Aviation Loan Terms;
1.1.22	Engines	 (a) each of the engines described in the Aircraft details whether or no from time to time installed on the Airframe or any other airframe but which, having been removed from the Airframe, remains the property of the Customer;
		(b) any other engine which may from time to time be installed upon o attached to the Airframe which becomes the property of the Customer;
		(c) in so far as the same belong to the Customer , any and al appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time relating to an engine referred to in (a) or (b) above whether or not installed on or attached to such engine; and
		(d) in so far as the same belongs to the Customer , all substitutions replacements or renewals from time to time made on or to an items referred to in (a), (b) and (c) above in accordance with this Deed :
1.1.23	Eurocontrol	the Central Route Charges Office, European Organisation for the Safety of Air Navigation;
1.1.24	Eurocontrol Letter	means the letter sent or to be sent by the Customer to Eurocontrol is respect of air navigation charges in relation to the Aircraft in the form se out in Schedule 4;
1.1.25	Event of Default	has the meaning given to it in the Aviation Loan Terms;
1.1.26	Finance Document	has the meaning given to it in the Aviation Loan Terms;
1.1.27	Financial Indebtedness	has the meaning given to it in the Aviation Loan Terms;
1.1.28	Guarantor	has the meaning given to it in the Aviation Loan Terms;
1.1.29	Holding Company	has the meaning given to it in the Aviation Loan Terms;
1.1.30	Insurance(s)	all policies or contracts of insurance relating to the Aircraft;
1.1.31	Insurers	in respect of the Insurances, such insurance underwriters and/or insurance companies in the major international insurance markets as may be, fror time to time, approved by Lombard in accordance with the terms of thi Deed;
1.1.32	Interest	has the meaning given to it in clause 2.1.1;
1.1.33	International Registry	the registry established pursuant to the Cape Town Convention an Aviation Protocol;
1.1.34	Maintenance Agreement	has the meaning given to it in the Aviation Loan Terms;
1.1.35	Maintenance Performer	has the meaning given to it in the Aviation Loan Terms;
1.1.36	Maintenance Programme	a programme for the maintenance of the Aircraft prepared and/or approve by the Aviation Authority and the manufacturer of the Aircraft;
1.1.37	Major Damage	has the meaning given to it in clause 5.2.2;

	· ·	
1.1.38	Major Damage Sum	the major damage sum specified at the front of this Deect;
1.1.39	Material Adverse Effect	a change in condition or circumstances (including any legal, tax or VAT treatment) which in Lombard's view is materially adverse to any Obligor or to the assets, business or financial condition or trading prospects of any Obligor such that the Obligor could reasonably be expected to be unlikely to meet any of its payment or other obligations under the under the Finance Documents or any other agreement with Lombard;
1.1.40	Obligor	the Customer, any Guarantor and any Security Provider;
1.1.41 1.1.42	Operating Agreement Operator	has the meaning given to it in the Aviation Loan Terms; such person operating, chartering or leasing the Aircraft as permitted under
		clause 9;
1.1.43	Parts	means, all modules, appliances, parts, accessories, auxiliary power unit, instruments, furnishings and other equipment of whatsoever nature which are from time to time attached to the Airframe or an Engine or which, having been removed therefrom, remains the property of the Customer ;
1.1.44	Permitted Lien	has the meaning given to it in the Aviation Loan Terms;
1.1.45	Permitted Transaction	has the meaning given to it in the Aviation Loan Terms;
1.1.46	Propellers	means:
		(a) each of the propellers described in the Aircraft details whether or not from time to time installed on the Airframe or any other airframe but which, having been removed from the Airframe, remains the property of the Customer; or
		(b) any other propeller which may from time to time be installed upon or attached to the Airframe which becomes the property of the Customer ; and
		(c) in so far as the same belong to the Customer , any and all appliances, instruments or accessories or other equipment or Parts of whatever nature from time to time relating to a propeller referred to in (a) or (b) above whether or not installed on or attached to such engine; and
		in so far as the same belongs to the Customer , all substitutions, replacements or renewals from time to time made on or to any items referred to in (a), (b) and (c) above in accordance with this Deed ;
1.1.47	Relevant Jurisdiction	has the meaning given to it in the Aviation Loan Terms;
1.1.48	Requisition	(a) any expropriation, confiscation, requisition or acquisition of the Aircraft, whether for full consideration, a consideration less than its proper value, a nominal consideration or without any consideration, which is effected by any government or official authority or by any person or persons claiming to be or to represent a government or official authority (excluding a requisition for hire for a fixed period not exceeding one year without any right to an extension); and / or
		(b) any arrest, capture, seizure, sequestration, forfeiture, compulsory acquisition or detention of the Aircraft (including any hijacking or theft or disappearance),
		in each case unless the Aircraft is redelivered to the full control of the relevant Obligor within thirty (30) days;
1.1.49	Requisition Compensation	all sums payable by reason of a Requisition and/or Compulsory Acquisition of the Aircraft;
	and the second	

1.1.50	Security Assets	the Aircraft and all assets which from time to time are subject to the security created or expressed to be created by this Deed ;
1.1.51	Security Documents	has the meaning given to it in the Aviation Loan Agreement;
1.1.52	Security Period	the period starting from the date of this Deed and ending on the date on which the Customer's Obligations have been unconditionally and irrevocably repaid and discharged in full and Lombard has no commitment to lend to the Customer ;
1.1.53	SPV	has the meaning given to it in the Aviation Loan Terms;
1.1.54	State of Registration	has the meaning given to it in the Aviation Loan Agreement;
1.1.55	Subsidiary	has the meaning given to it in the Aviation Loan Terms;
1.1.56	Tax	has the meaning given to it in the Aviation Loan Terms;
1.1.57	Technical Records	means all records, logs, technical data and other materials and documentation relating to the Aircraft and the maintenance and operation thereof, including, without limitation, those which are from time to time required to comply with the regulations and requirements of the State of Registration or which are kept in purported compliance with such regulations and requirements;
1.1.58	Total Loss	has the meaning given to it in the Aviation Loan Terms;
1.1.59	VAT	has the meaning given to it in the Aviation Loan Terms; and
1.1.60	Warranties	any agreement under which warranties are provided in favour of the Customer in relation to the Aircraft.
1.2 1.3	defined in this Deed .	.oan Agreement have the same meaning in this Deed unless expressly riation Loan Agreement apply to this Deed as though they were set out in
1.4	full in this Deed , except that references to the Unless a contrary indication appears a referen	e Aviation Loan Agreement will be construed as references to this Deed. nce in this Deed to "Customer's Obligations" where the Customer is more at, several and independent liabilities and references to "Customer" are to
1.5	It is intended that this document takes effect	as a deed notwithstanding that a party may only execute it under hand.
2	Customer's Obligations	
2.1	The Customer agrees to pay to Lombard on all the Customer's liabilities to Lombard (pri another) including (without limitation):	demand all the Customer's Obligations. The Customer's Obligations are esent, future, actual or contingent and whether incurred alone or jointly with
2.1.1	interest at the rate charged by Lombard in ac before and after demand or judgment or deci	ccordance with the terms of the Aviation Loan Agreement, calculated both ree on a daily basis ("Interest"); and
2.1.2	from the date of payment) in connection wi	es, Lombard or a receiver incurs (on a full indemnity basis and with Interest th the Aircraft or taking, perfecting, protecting or enforcing this Deed or Lombard will provide details of its costs and expenses to the Customer.
2.2	Interest due and unpaid will be compounded and payable.	monthly on the days selected by Lombard but will remain immediately due
2.3	be made in full, without set-off or deduction required by law in which event the Custome	ents under this Deed is absolute and unconditional and each payment must , or withholding for any Taxes , save for any withholding Taxes which are or shall pay to Lombard such additional amount as is necessary to ensure equal to the amount Lombard would have received in the absence of any tion or withholding.
3	Charge	en 1935 Produktion (1930) - Andrea Martines, andreas (1930) - Andreas (1930) - Andreas (1930)

3.1	As a continuing security for the payment on demand of the Customer's Obligations the Customer hereby unconditionally and irrevocably and with full title guarantee:
3.1.1	by way of first priority legal mortgage, mortgages its respective interest in the Aircraft to Lombard;
3.1.2	assigns to Lombard absolutely by way of security all of its right, title, benefit and interest, now or at any later time, in and to:
	(a) all Charters, Earnings and any Charter Guarantee;
	(b) all Insurances and proceeds of the Insurances;
	(c) all Warranties, Operating Agreements and Maintenance Agreements;
	(d) all Requisition Compensation; and
	(e) any sale and purchase agreement which the Customer is party to in relation to the Aircraft.
3.2	The assignment under clause 3.1.2 above is subject to a proviso for reassignment on irrevocable discharge in full of the Customer's Obligations . Following the expiry of the Security Period Lombard shall, at the request and cost of the Customer promptly execute and deliver to the Customer a discharge of this Deed together with all further instruments and documents which the Customer may require for the purpose of discharging the security constituted by this Deed .
3.3	If any property expressed to be assigned in this clause 3 cannot be assigned, the Customer shall hold that property on trust for Lombard and the assignment shall take effect as an assignment of any damages, compensation, remuneration, profit, rent or income which the Customer may derive from that right or be entitled to in respect of that right.
3.4	Any Part or Engine which becomes the property of the Customer pursuant to clause 8.2.5 shall, without the need for any further act, become subject to this Deed and shall form part of the Security Assets.
4	Perfection of security
4.1	The Customer shall promptly on request by Lombard, give notice of assignment:
4.1.1	in form and substance satisfactory to Lombard, to each party with whom the Customer has entered into a Charter, Charter Guarantee, Operating Agreement or Maintenance Agreement or any other agreement under which Earnings are paid;
4.1.2	substantially in the form set out in Schedule 2, to each Insurer;
4.1.3	substantially in the form set out in Schedule 3, to each person who has given Warranties or entered into a Charter, Charter Guarantee, Operating Agreement or Maintenance Agreement in relation to the Aircraft,
	and shall use all reasonable endeavours to procure that each addressee of such notice will promptly provide an acknowledgement to Lombard in form and substance satisfactory to Lombard.
5	r: Receipts
5.1	The Earnings and any sums recoverable under any Charter Guarantee, Operating Agreement or Maintenance Agreement shall be payable to the Customer until Lombard directs to the contrary following the occurrence of an Event of Default, on and after which direction:
5.1.1	the Customer shall immediately pay such amounts to Lombard or as Lombard may direct, and Lombard may instruct all persons from whom those sums are due to do the same; and
5.1.2	any such sums then held by the Customer's brokers, bankers or other agents or representatives shall be deemed to have been received by, and to be held by them, on trust for Lombard .
5.2	Before an Event of Default occurs, sums recoverable in respect of the Insurances shall be payable as follows:
5.2.1	any sum recoverable in respect of a Total Loss under the Insurances shall be paid to Lombard;
5.2.2	any sum recoverable under the Insurances in respect of a casualty to the Aircraft in respect of which the claim or the aggregate of the claims against all insurers (before adjustment for any relevant franchise or deductible) exceeds the Major Damage Sum ("Major Damage") shall be paid to Lombard save that:

	(a) the sum received by Lombard shall be paid over to the Customer upon the Customer providing evidence satisfactory to Lombard that all loss and damage resulting from the casualty has been properly made good and repaired and that liabilities connected with the casualty have been paid by the Customer; and
	(b) the insurers with whom the Insurances are effected may in the case of any Major Damage , and with the prior written consent of Lombard , make payment to the repairers on account of the repairs which are being carried out;
5.2.3	any sum recoverable in respect of a Total Loss of an Engine shall be paid to Lombard save that where Lombard has accepted a replacement engine for the Engine which has suffered a Total Loss (and such replacement engine has become an Engine for the purposes of this Deed) the sum received by Lombard shall be paid over to the Customer or to the seller of that replacement Engine ;
5.2.4	any other sum recoverable under the Insurances shall be paid to the Customer which shall apply it in making good the loss and fully repairing all damage in respect of which that insurance money was received; and
5.2.5	any sum recoverable under the Insurances against general third party liability shall be paid directly to the person to whom the Customer is liable or to the Customer in reimbursement of moneys expended by it to discharge that liability.
5.3	On or after the occurrence of an Event of Default, any sums recoverable under the Insurances (other than those referred to in clause 5.2.5) shall be payable to Lombard.
5.4	On or after the occurrence of an Event of Default, any sums recoverable in respect of the Warranties shall be payable to Lombard.
6	Representations and warranties
6.1	The Customer represents and warrants to Lombard on the date it signs this Deed that:
6.1.1	it is a limited liability company or a limited liability partnership duly incorporated or established under the laws of the Jurisdiction of Incorporation and it has the power to carry on its business and to own its property and other assets;
6.1.2	the obligations expressed to be assumed by it in each Finance Document to which it is a party are legal, valid, binding and enforceable obligations;
6.1.3	it has power to execute and perform its obligations under those Finance Documents to which it is a party;
6.1.4	the execution and performance by it of the Finance Documents to which it is a party does not:
	(a) contravene any applicable law, order or regulation;
	(b) conflict with, or result in any breach of any of the terms of, or constitute a default under, any agreement or other instrument to which it is a party or any licence or other authorisation to which it is subject or by which it or any of its property is bound unless such conflict, breach or default could not reasonably be expected to have a Material Adverse Effect ; or
	(c) contravene or conflict with its memorandum, articles of association or any other constitutional documents relating to it;
6.1.5	it has and will maintain all necessary Authorisations to perform its obligations under the Finance Documents;
6.1.6	any Authorisation required or desirable to make any Finance Document admissible in evidence in each Relevant Jurisdiction has been obtained or effected and is in full force and effect;
6.1.7	the choice of governing law of any Finance Document to which it is a party will be recognised and enforced in each Relevant Jurisdiction;
6.1.8	any judgment obtained in relation to any Finance Document to which it is a party in the jurisdiction of the governing law of that Finance Document will be recognised and enforced in each Relevant Jurisdiction;
6.1.9	any filing, recording or enrolment with any court or other authority of any Finance Document to which it is a party necessary under the laws of any Relevant Jurisdiction has been completed;
6.1.10	any stamp, registration, notarial or similar Tax or fee to be paid (on or in relation to):
an An An Spearsean	(a) any Finance Document to which it is a party; or
a di kasa sha	

- (b) the transactions contemplated by any Finance Document to which it is a party; or
- (c) under the laws of any Relevant Jurisdiction,

(b)

has been paid or will be paid by the time required by law;

- 6.1.11 it is not necessary under the laws of any Relevant Jurisdiction:
 - (a) in order to enable Lombard to enforce its rights under any Finance Document to which the Customer is a party; or
 - by reason of the execution of any Finance Document to which the Customer is a party or the performance by Lombard of its obligations under any Finance Document to which the Customer is a party.
 - that Lombard should be licensed, qualified or otherwise entitled to carry on business in any Relevant Jurisdiction;
- 6.1.12 **Lombard** will not be deemed to be resident, domiciled or carrying on business in any **Relevant Jurisdiction** by reason only of the execution, performance and/or enforcement of any **Finance Document** to which the **Customer** is a party;
- 6.1.13 it has not taken any action nor have any steps been taken or legal proceedings been started or to the best of its knowledge threatened in writing against it for:

(a) winding-up, dissolution, re-organisation, voluntary arrangement under Part I of the Insolvency Act 1986 or any other scheme of arrangement, restructuring plan, the granting of a moratorium or a compromise or composition with any of its creditors; or

- (b) the enforcement of any Encumbrance over its assets; or
- (c) the appointment of a receiver, administrative receiver, administrator, trustee or similar officer of it or of any of its assets;
- 6.1.14 except as disclosed to Lombard in writing before the date of signing this Deed, no action, litigation, arbitration or administrative proceeding has been commenced or to the best of its knowledge is pending or threatened against it or any member of the Customer (where the Customer is a LLP) or any Subsidiary of the Customer, involving a potential liability which could if adversely determined, individually or in aggregate, have a Material Adverse Effect nor is there subsisting any unsatisfied judgment or award given against it or any such member or Subsidiary;

6.1.15 except as disclosed to Lombard in writing before the date of signing this Deed, to the best of its knowledge, having made reasonable enquiry, no action, litigation, arbitration or administrative proceeding has been commenced or is pending or threatened against any Obligor (other than the Customer) or any Holding Company or member of such Obligor or Holding Company (where such Obligor or Holding Company is a LLP) involving a potential liability which could, individually or in aggregate, have a Material Adverse Effect nor is there subsisting any unsatisfied judgment or award given against any such Obligor, Holding Company or member;

- 6.1.16 it has not made any representations or warranties, whether express or implied, in connection with the Finance **Documents** which may increase **Lombard's** obligations or adversely affect **Lombard's** rights;
- 6.1.17 all information supplied by it or by any person on its behalf to **Lombard** was and remains true and complete in all respects and it is not aware of any material fact or circumstance which has not been disclosed to **Lombard**;
- 6.1.18 the Aircraft is not subject to any Encumbrance other than any Permitted Lien;
- 6.1.19 where it is stated or shown in the Deed that the Customer is a SPV:
 - (a) that no Encumbrance exists over all or any of its assets other than any Permitted Lien; and
 - (b) it has no Financial Indebtedness outstanding other than a Permitted Transaction;
- 6.1.20 the security created by each Security Document has or will have the ranking which it is expressed to have in the Security Document and it is not subject to any prior ranking or pari passu ranking Encumbrance other than as permitted in writing by Lombard or as exists by operation of applicable law;
- 6.1.21 it has a good, valid and marketable title to, or valid leases or licences of, and all appropriate Authorisations to use, the assets necessary to carry on its business as presently conducted;
- 6.1.22 it is the sole legal and beneficial owner of the assets over which it purports to grant security pursuant to the Security Documents to which it is a party including, without limitation, the Aircraft; and

6.1.23	where it is stated in this Deed, that the Aircraft shall be used in the course of the Customer's business only and not otherwise.
6.2	Each of the representations set out above are repeated by the Customer on the date of this Deed and the last day of each Charging Period by reference to the facts and circumstances then existing.
7	Aircraft undertakings
7.1	The Customer will at all times during the Security Period (or, if applicable, shall procure that any Operator of the Aircraft will):
7.1.1	register and keep the Aircraft registered at the Aviation Authority with the same registration mark either:
na series Terres de la composición Terres de la composición de la composición de la composición de la composición de la c	(a) in the Customer's name as owner; or
	(b) if applicable, in the name of the Operator,
	in accordance with all applicable laws and regulations;
7.1.2	register this Deed and keep this Deed registered against the Aircraft in the register of aircraft mortgages maintained by the Aviation Authority;
7.1.3	not do, and shall procure that no third party does, any act or thing which might prejudice or cancel the registrations referred to in clauses 7.1.1 and 7.1.2;
7.1.4	inform Lombard immediately of any requirement imposed by any insurer of the Aircraft or an authority with which the Customer is unable to comply;
7.1.5	use the Aircraft in a safe, careful and proper manner and in accordance with the manufacturer's recommendations;
7.1.6	provide to Lombard annually copies of any maintenance records, log books and insurance information relating to the Aircraft;
7.1.7	ensure the Aircraft is operated in accordance with all applicable laws and regulations (including those of the Aviation Authority) and in a manner which does not prejudice the rights of Lombard in respect of the Security Assets, not engage in any unlawful trade or allow the Aircraft to carry a cargo which will expose the Aircraft to penalty or forfeiture;
7.1.8	pay when due and discharge or cause to be promptly paid and discharged as they fall due all costs and expenses of operating, storing and maintaining the Aircraft and performing its obligations under the documents in relation to the Aircraft (including all taxes, fines, licence fees, charges of airport authorities (whether relating to landing fees, parking fees, handling charges or otherwise), all charges imposed by air navigation authorities (including Eurocontrol) and all charges of the Aviation Authority (whether relating to navigation or otherwise)) and will on written request by Lombard , supply evidence of the payment of any such charges;
7.1.9	immediately inform Lombard of any arrest, proceedings, collision, incident, or damage (where such damage exceeds the Major Damage Sum) or Total Loss affecting the Alrcraft;
7.1.10	not, without the prior written consent of Lombard, change the Habitual Base from that notified to Lombard prior to the date of this Deed. This clause shall not apply where the Habitual Base is changed to a location in England provided always Lombard is given prior written notice of such location;
7.1.11	permit Lombard and its authorised representatives to inspect the Aircraft at any time upon giving reasonable notice and for that purpose grant or use all reasonable endeavours to procure access for Lombard and its authorised representatives to any land or buildings on or in which the Aircraft or any Engine or Part may be stored provided however that any such inspection shall not (unless an Event of Default or a Default has occurred) disturb the normal commercial operation or maintenance of the Aircraft;
7.1.12	not allow the Aircraft to be used in any part of the world where there are hostilities (whether war is declared or not) or enter a zone which is declared a war zone by any government or by the Insurers;
7.1.13	ensure that where the Customer is not operating the Aircraft, that the operation of the Aircraft is governed by an Operating Agreement and where required by Lombard registered with the International Registry;
7.1.14	document any commercial use of the Aircraft on market standard terms;
7.1.15	perform and discharge all duties, liabilities or obligations imposed on the Customer under any Charter, Operating Agreement or other contract relating to the Aircraft;

 7.1.16 if requested by Lombard, deposit with Lombard all deeds and documents of title, leases or Charters relating t Security Assets other than those documents that by operation of applicable law must remain on the Aircraft; 7.1.17 not create or allow to subsist, or permit or permit any other person to create or grant, any Encumbrance over the Sec Assets or any part thereof and shall procure the immediate discharge of any Encumbrance which arises or exist respect of the Security Assets or any part thereof other than: (a) a Permitted Lien in respect of the Aircraft; or (b) any other Encumbrance disclosed to Lombard in writing prior to the date of the Aviation Loan Agreet and confirmed in the written confirmation provided under the Aviation Loan Terms; 7.1.18 not assign or purport to assign any Earnings or Insurances in relation to the Aircraft other than under the Fin Documents; 7.1.19 not amend or end, or grant a waiver under, any contract mortgaged, charged, assigned or held on trust under this E 7.1.20 promptly, upon becoming aware of the same, notify Lombard of any Encumbrance over the Aircraft; 7.1.21 if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fire metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first primortgage in favour of Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire metal plate which has inscribed thereon (in legible characters) the following statement: 	
 Assets or any part thereof and shall procure the immediate discharge of any Encumbrance which arises or exist respect of the Security Assets or any part thereof other than: (a) a Permitted Lien in respect of the Aircraft; or (b) any other Encumbrance disclosed to Lombard in writing prior to the date of the Aviation Loan Agreed and confirmed in the written confirmation provided under the Aviation Loan Terms; 7.1.18 not assign or purport to assign any Earnings or Insurances in relation to the Aircraft other than under the Fin Documents; 7.1.19 not amend or end, or grant a waiver under, any contract mortgaged, charged, assigned or held on trust under this E 7.1.20 promptly, upon becoming aware of the same, notify Lombard of any Encumbrance over the Aircraft; 7.1.21 if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fire metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first pr mortgage in favour of Lombard "; 7.1.22 if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire 	o the
 (b) any other Encumbrance disclosed to Lombard in writing prior to the date of the Aviation Loan Agreed and confirmed in the written confirmation provided under the Aviation Loan Terms; 7.1.18 not assign or purport to assign any Earnings or Insurances in relation to the Aircraft other than under the Fin Documents; 7.1.19 not amend or end, or grant a waiver under, any contract mortgaged, charged, assigned or held on trust under this E 7.1.20 promptly, upon becoming aware of the same, notify Lombard of any Encumbrance over the Aircraft; 7.1.21 if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fire metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first primortgage in favour of Lombard "; 7.1.22 if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire 	urity sts in
 and confirmed in the written confirmation provided under the Aviation Loan Terms; 7.1.18 not assign or purport to assign any Earnings or Insurances in relation to the Aircraft other than under the Fin Documents; 7.1.19 not amend or end, or grant a waiver under, any contract mortgaged, charged, assigned or held on trust under this E 7.1.20 promptly, upon becoming aware of the same, notify Lombard of any Encumbrance over the Aircraft; 7.1.21 if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fire metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first pr mortgage in favour of Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire 	ائي دريان دران دريور
Documents; 7.1.19 not amend or end, or grant a waiver under, any contract mortgaged, charged, assigned or held on trust under this E 7.1.20 promptly, upon becoming aware of the same, notify Lombard of any Encumbrance over the Aircraft; 7.1.21 if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fire metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first pr mortgage in favour of Lombard "; 7.1.22 if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire	ment
 7.1.20 promptly, upon becoming aware of the same, notify Lombard of any Encumbrance over the Aircraft; 7.1.21 if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fire metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first pr mortgage in favour of Lombard "; 7.1.22 if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire 	ance
 7.1.21 if requested by Lombard, promptly affix and keep affixed to the Aircraft in a prominent position in the cockpit a fire metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first pr mortgage in favour of Lombard "; 7.1.22 if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire)eed;
metal plate which has inscribed thereon (in legible characters) the following statement: "This Aircraft, its engines and accessories are owned by TAG Aviation (Stansted) Ltd and subject to a first pr mortgage in favour of Lombard "; 7.1.22 if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire	
mortgage in favour of Lombard "; 7.1.22 if requested by Lombard, promptly affix and keep affixed to each of the Engines underneath the cowling a fire	proof
	iority
	proof
"This Engine is owned by TAG Aviation (Stansted) Ltd and subject to a first priority mortgage in favour of Lombard	•
7.1.23 not sell, transfer or otherwise dispose of the Aircraft or any part thereof or interest therein other than with the prior w consent of Lombard (which consent Lombard shall have absolute discretion to withhold);	ritten
7.1.24 promptly give written notice to Lombard of the occurrence of any Event of Default, Default, Total Loss or any cond event or circumstance which, with the lapse of time, might constitute a Total Loss;	ition,
7.1.25 comply with any other requirements which Lombard notifies to the Customer;	
7.1.26 retain a copy of any certificate issued by the relevant tax authorities evidencing that VAT has been paid in respect of Aircraft on board the Aircraft;	of the
7.1.27 if VAT has not been paid in respect of the purchase of the Aircraft by the Customer, ensure that VAT does not been payable in respect of the Aircraft;	ome
7.1.28 take (at its own cost) all steps required from time to time by applicable law or by customary and established practices as Lombard may reasonably request to protect and perfect Lombard's interest in the Aircraft as mortgagee;	ce or
7.1.29 procure that the Aircraft:	
(a) on and at all times following the date falling six weeks after drawdown under the Aviation Loan Agreement if earlier, the date on which the Aviation Authority first issues a valid certificate of airworthiness for the Airco possesses a valid certificate of airworthiness issued by the Aviation Authority;	nt (or raft),
(b) possess a valid certificate of registration issued by the Avlation Authority;	
(c) possess (if issued by the Aviation Authority) a valid annual airworthiness review certificate;	
(d) possesses a valid annual release to service issued by the Maintenance Performer; and	
(e) possesses a valid Air Operator's Certificate on which the Aircraft is listed where the Aircraft is use commercial air transport or aerial work.	d for
7.1.30 deliver a copy of each renewed or reissued certificate of airworthiness, certificate of registration, airworthiness recertificate and a valid annual release to service to Lombard promptly upon such renewal or each annual reissue of certificate (if applicable) by the Aviation Authority;	

7.1.31	not, remove or permit the removal of any Part or any equipment belonging to the Aircraft (other than in accordance with clause 8.2) without Lombard's prior written consent;
7.1.32	procure that no steps are taken to deregister the Aircraft from the register of aircraft, or this Deed from the register of mortgages, maintained by the Aviation Authority without the prior written consent of Lombard (which consent Lombard shall have absolute discretion to withhold);
7.1.33	procure that the Aircraft is at all times insured in accordance with the requirements of clause 10; and
7.1.34	following an Event of Default which is continuing the Customer shall forthwith, upon the request of Lombard but at no cost to Lombard (and without prejudice to the rights of Lombard pursuant to the Deregistration Power of Attorney), execute all such documents or agreements and take all such steps as are within its power and as are required in order to de-register the Aircraft from registration with, and export the Aircraft from, the State of Registration.
8	Operational and maintenance undertakings
8.1	The Customer will at all times during the Security Period (or, if applicable, shall procure that any Operator of the Aircraft will):
8.1.1	keep the Aircraft in good and airworthy condition and maintain and repair the Aircraft, the Engines and all Parts in accordance with (i) the Maintenance Programme, (ii) all mandatory rules and regulations of the Aviation Authority to the standard required by the Aviation Authority and (iii) any other regulations or requirements necessary in order to enable the Aircraft at any time to obtain a certificate of airworthiness issued by the Aviation Authority plus a valid annual release to service stamp issued by the Maintenance Performer;
8.1.2	maintain or cause to be maintained (in writing and in English) all logbooks, maintenance records, record books and other documents as are required by the Aviation Authority to the standard required by the Aviation Authority or the Maintenance Programme to be maintained with respect to the Aircraft;
8.1.3	replace or cause to be replaced promptly any Part of the Aircraft or any Engine in respect of which a Total Loss occurs with a replacement Part or Engine which shall be (i) owned by the Customer free and clear of all Encumbrances other than Permitted Liens (ii) subject to the Encumbrances intended to be created by this Deed and (iii) in such operating condition as, and of equivalent or better value to, the replaced Part or Engine was (or would have been if it had been maintained in accordance with the terms hereof), and where a Part or Engine is replaced under this clause, the Customer agrees to enter into a mortgage supplement in respect of such replacement Part or Engine;
8.1.4	make or cause to be made all mandatory or recommended alterations and modifications and additions to the Aircraft and the Engines as may from time to time be required by the Aviation Authority within the time period required by the Aviation Authority or any other relevant authority or the manufacturer of the Aircraft or any Engine;
8.1.5	comply with all mandatory Airworthiness Directives from time to time issued by the Aviation Authority or any other relevant authority and any modification recommended by the manufacturer of the Aircraft or any Engine in a mandatory service bulletin (unless such mandatory service bulletin is excluded by the Aviation Authority);
8.1.6	ensure that all inspections (in connection with maintenance), services, repairs, overhauls, tests and modifications to be made or carried out to or on the Alrcraft are made or carried out by a maintenance performer approved by the Aviation Authority;
8.1.7	not suffer any arrest, detention, exercise of a possessory lien, execution, distress or other legal process to be levied or sued out against the Aircraft and in the event of an arrest, detention, exercise of a possessory lien, execution, distress or other legal process as aforesaid, the Customer shall notify Lombard immediately and shall take all steps necessary to ensure or procure that the Aircraft is promptly released therefrom;
8.1.8	ensure that such crew and engineers as are employed in connection with the operation and maintenance of the Aircraft have such qualifications and hold such licences as are required by the Aviation Authority and by all other applicable laws and regulations;
8.1.9	operate the Aircraft or cause the Aircraft to be operated only in the ordinary course of its business and not operate or permit the Aircraft to be operated for testing (other than for maintenance), training (other than of in-house pilots), racing, experimental flying or any other purpose for which it is not designated or reasonably suitable;
8.1.10	at any time upon giving reasonable notice make available to Lombard for inspection the Technical Records and other log books of the Aircraft and all other repair, maintenance, modification, overhaul or other records kept by the Customer or any other person in respect of the Aircraft; and
8.1.11	on request, promptly furnish to Lombard all such information regarding the operation, maintenance and condition of the Aircraft as Lombard may reasonably request.

2	Until the occurrence of an Event of Default, the Customer may:
8.2.1	remove any Part (other than an Engine) in the ordinary course of maintenance, service, repair, overhaul or testing, if is promptly replaces it with a replacement Part which satisfies the requirements of clause 8.1.3;
8.2.2	remove any Engine in the ordinary course of maintenance, service, repair, overhaul or testing provided that such Engine is re-installed on the Aircraft, or a replacement engine which satisfies the requirements of clause 8.1.3 is installed or the Aircraft, within thirty days (or such longer period as may be permitted by Lombard) of the removal from the Aircraft of such Engine and provided further that the removed Engine remains the property of the Customer unless and until a replacement engine satisfying the requirements of clause 8.1.3 is installed on the Aircraft;
8.2.3	make such alterations and modifications in and additions to the Aircraft as it thinks desirable in the proper conduct of its business or which may arise as a result of an Airworthiness Directive or a service bulletin provided that no such alteration, modification or addition (other than an Airworthiness Directive or mandatory service bulletin) breaches the requirements or recommendations of the Aviation Authority, the Maintenance Programme, the manufacturer or any other relevant authority or materially diminishes the value or utility of the Aircraft below that prevailing immediately prior to such modification or alteration or impairs the airworthiness of the Aircraft and further provided that no such alteration or modification may be made with or comprising Parts which are not in the absolute legal and beneficial ownership of the Customer free from all Encumbrances other than Permitted Liens;
8.2.4	remove any Part (other than an Engine) from the Aircraft without replacing it as required pursuant to the terms hereof if such Part (i) is in addition to and not in replacement of any Part previously installed on or attached to the Aircraft, (ii) is not required to be installed on or attached to the Aircraft by the terms hereof and (iii) can be removed from the Aircraft without impairing the airworthiness of the Aircraft or materially diminishing the value of the Aircraft below the value which the Aircraft would have had had such additional Part not been installed thereon or attached thereto;
8.2.5	whilst an Engine has been removed from the Aircraft in accordance with the provisions of clause 8.2.2, install a replacement engine notwithstanding that such replacement engine is not the property of the Customer provided that the removed Engine is re-installed, or such other engine as may satisfy the requirements of clause 8.1.3 is installed, on the Aircraft within the period specified in clause 8.2.2 and, further provided that the Engine substituted or replaced by such replacement engine shall at all times remain free from any Encumbrance other than Permitted Liens, shall remain the property of the Customer, and shall not cease to be subject to this Deed unless and until it is replaced by an engine which satisfies the requirements of clause 8.1.3; and
8.2.6	print or paint its name or other symbol on the Aircraft.
8.3	Save insofar as expressly required or permitted by the terms hereof and subject in any case as provided in clause 8.2, the Customer shall not, without the prior written consent of Lombard make any alteration in or modification or addition to the Aircraft , remove any part from the Aircraft or install any Part on or attach any Part to the Aircraft or permit any of the said events to occur.
8.4	The Customer shall promptly notify Lombard:
8.4.1	of any material alterations in or material modifications or additions to the Aircraft;
8.4.2	of any Encumbrance (excluding Permitted Liens) arising on the Aircraft and (in the case of all liens including Permitted Liens) exercised over the Aircraft or any part thereof;
8.4.3	(if Lombard so requests) the location of the Aircraft and of any Engine which is not, for the time being, installed on or attached to the Aircraft.
8.5	The existence of any agreement for maintenance of the Aircraft shall not in any way absolve, negate, reduce or mitigate the Customer's obligations to maintain the Aircraft in the manner required under this Deed.
8.6	The Customer shall:
8.6.1	ensure that the Aircraft shall not be used or operated when the Insurances are not in full force and effect or otherwise at a time or in a manner or in any geographical area which constitutes a breach of the terms of the Insurances or is otherwise inconsistent with the terms of the Insurances;
8.6.2	not cause or permit the Aircraft to be operated in contravention of any airworthiness certificate or the Aircraft operating manual relating thereto for the time being;
8.6.3	not make, do, permit, consent or agree to any act or omission to act (save as permitted or contemplated by this Deed) which would or might cause the Aircraft, or any Part thereof, to be exposed to penalty, forfeiture, impounding, destruction

8.6.4	not hold out or represent or permit the holding out or representation that any member of the crew of the Aircraft is in the employment of Lombard or that Lombard is carrying persons or goods in the Aircraft or are in any way connected or associated (whether for hire or gratuitously) with any operation of the Aircraft undertaken (or on behalf of) by the
	Customer;
8.6.5	not pledge the credit of Lombard for any maintenance, service or repairs to, or replacement or overhaul, of the Aircraft or any Engine or Part; and
8.6.6	pay and discharge, or cause to be paid and discharged, when due and payable, or make adequate provision by way of security or otherwise for all debts, damages, claims and liabilities which may have given or may reasonably be expected to give rise to any Encumbrance (other than a Permitted Lien) over or affecting the Aircraft.
9	Possession, subleasing and chartering
9.1	The Customer may not, without Lombard's prior written consent, charter the Aircraft.
9.2	The Customer may deliver possession of the Aircraft, Airframe, any Engine or any Part to the relevant Maintenance Performer or to any other suitably qualified maintenance contractor approved by Lombard for the purpose of performing maintenance or other work required or permitted by this Deed.
9.3	Save as permitted by clauses 9.1 and 9.2, the Customer shall not lease, or otherwise part with possession of, the Alrcraft without the prior written consent of Lombard, which consent Lombard may grant or withhold in its absolute discretion.
9.4	If, pursuant to clause 9.1 or 9.3 Lombard consents to the Aircraft being leased or subleased, or otherwise operated by any other person, the Customer shall procure that:
9.4.1	the rights of that person under that lease, sublease or other operating agreement are expressed to be subject and subordinate to the rights of Lombard under this Deed;
9.4.2	that person executes and delivers to Lombard (prior to the commencement of that lease, sublease or other operating agreement) a Deregistration Power of Attorney, an Authority Letter and a consent and subordination agreement in terms specified by Lombard;
9.4.3	the lease, sublease or other operating agreement contains provisions which are at least as onerous as those set out in clauses 8 and 10;
9.4.4	the lease, sublease or other operating agreement contains such terms that shall not cause the Customer to be in breach of any Finance Document or cause an Event of Default;
9.4.5	the rights under that lease, sublease or other operating agreement (and any rights in relation to the insurances and warranties relating to the Aircraft, the Airframe or any Engine or Part) are assigned to Lombard (prior to the commencement of that lease, sublease or other operating agreement) in terms specified by Lombard.
10	Insurance undertakings
10.1	The Customer will at all times during the Security Period effect and maintain (or cause to be effected and maintained) at its own expense in relation to the Aircraft:
10.1.1	"Hull All Risks" insurance for loss of or damage to the Aircraft whilst flying and on the ground on an agreed value basis in an amount in the Required Insurance Currency which is no less than the Required Insurance Value;
10.1.2	"Spares All Risks" insurance (including "Hull, War and Allied Perils Risk" to the fullest extent available) on the Engines and Parts while not installed on the Airframe (and not considered part of the aircraft for insurance purposes) of the Aircraft for their full replacement cost (unless otherwise insured on an agreed value basis) in the Required Insurance Currency;
10.1.3	"Hull, War and Allied Perils" insurance, being such risks excluded from the "Hull All Risks" policy by application of War, Hijacking and Other Perils Exclusion Clause AVN48B (or subsequent version) to the fullest extent available from the leading aviation insurance markets, including confiscation and requisition by the State of Registration on an "agreed value basis" for an amount in the Required Insurance Currency which is no less than the Required Insurance Value .
10.2	The insurances required under clause 10.1 shall be provided on an agreed value basis (that is to say on the basis that in the event of a total loss the Insurers will pay the specified agreed value rather than the cost of acquiring a replacement aircraft) or as otherwise set out in this Deed and the policies shall include the form of AVN67B endorsement and, to the extent not in conflict with AVN67B:
10.2.1	include Lombard and its successors and assigns as additional insured and contract party for its rights and interests;

10.2.2 include a loss payable clause that provides that all insurance proceeds in respect of a Total Loss shall be payable to Lombard: 10.2.3 include a reference to the Aviation Loan Agreement and this Deed; 10.2.4 include, in the event of separate insurances being arranged to cover the "Hull All Risks" insurance and the "Hull War Risk" and related insurance, a 50/50 claims funding arrangement in the event of any dispute as to whether a claim is covered by the "Hull All Risks" or "Hull War Risks" policy, wording in the terms of Lloyds clause AVS103 being sufficient for such purpose; and 10.2.5 be subject to a deductible (except in the event of Total Loss) of not more than the Permitted Deductible. 10.3 The Customer shall effect and maintain (or cause to be effected and maintained) at its own expense in relation to the Aircraft, a policy or policies of insurance covering aircraft third party, passenger, baggage, cargo, mail and airline general third party (including products, premises and hangerkeepers) legal liability for a combined single limit (bodily injury/property damage) of not less than the Minimum Liability Cover (and in any event no less than the amount carried in respect of the same or similar aircraft in the Customer's fleet in the same jurisdiction as the Aircraft). War and Allied Perils Risks shall also be covered in accordance with the then current market practice for aircraft of the same type as the Aircraft. 10.4 The policies evidencing the insurances required under clause 10.3 shall: include Lombard and its officers, directors, employees, agents and its successors and assigns each as additional insured 10.4.1 and as contract parties; 10.4.2 not conflict with AVN 67B: 10.4.3 provide that all the provisions thereof, except the limits of liability, shall operate to give each additional insured the same protection as if there were a separate policy covering each named insured save that this provision shall not operate to include any claim howsoever arising in respect of loss or damage to the Aircraft or relevant spare parts covered under the Hull or spares insurance, it being understood that notwithstanding the foregoing the total liability of Insurers in respect of any and all insured parties shall not exceed the limits of liability stated in the policy; and 10.4.4 be primary and without right of contribution from other insurance which may be available to Lombard or any of its officers, directors, employees, agents and its successors and assigns. 10.5 The policies evidencing any Insurances with respect to the Aircraft required under clauses 10.1 and 10.3 shall: provide that the cover afforded to the additional insured shall not be invalidated by any act or omission (including 10.5.1 misrepresentation and non-disclosure) of any other person or party which results in a breach of any term, condition or warranty of the policy provided that the additional insured so protected has not caused, contributed to or knowingly condoned the said act or omission; 10.5.2 provide for worldwide coverage (subject only to such exceptions as are consistent with generally accepted market practice from time to time or such other exceptions as Lombard may reasonably agree in writing); provide that the Insurers shall waive any right of subrogation to any right against Lombard, any of its officers, directors, 10.5.3 employees, agents and its successors, transferees and assigns; 10.5.4 provide that Lombard shall not be liable for any premiums in respect thereof and that the Insurers shall not exercise any right of set-off or counterclaim in respect of unpaid premiums or otherwise against the interest of Lombard otherwise than in respect of unpaid premiums with respect to the Insurances of the Aircraft; and 10.5.5 provide that the required insurances may only be cancelled or materially altered in a manner adverse to the additional insureds and contract parties by the giving of not less than thirty days' notice in writing (but seven days or such lesser period as may be customarily available in respect of War and Allied Perils) to Lombard. Notice shall be deemed to commence from the date such notice is given by the Insurers. It is understood that such notice will not, however, be given at normal expiry date of the policy or any endorsement. All Insurances required under clauses 10.1 or 10.3 shall be maintained with such insurance underwriters and/or 10.6 insurance companies as Lombard may reasonably approve. 10.7 The Customer shall, in relation to the Aircraft, at its own cost and expense: furnish or procure that there is furnished to Lombard, forthwith upon the date not later than seven days after the renewal 10.7.1 date of each Insurance, a certificate or certificates, as hereinafter described, signed by the relevant Insurers (or their authorised representatives) or brokers, reasonably acceptable to Lombard, which have been appointed by the Customer (the "Brokers") providing evidence of insurance coverage pursuant to this clause 10. Each such certificate

shall (i) describe the perils covered by each policy of insurance issued by it then in force, (ii) specify the amounts of insurance coverage provided against each such peril and any deductible or uninsured amount and (iii) certify that each such policy is then in full force and effect; furnish or procure that there is furnished to Lombard, on written request, evidence of (a) payment of each sum then due 10.7.2 and payable under or in connection with any required Insurance and (b) before or on the expiry of any policy or policies in respect of such Insurances, renewal of the relevant policy or policies or that renewal will be effected on expiry; and 10.7.3 procure that the relevant Insurers or Brokers provide Lombard with such other information as to any Insurances as Lombard may reasonably request. 10.8 The Customer shall, in relation to the Aircraft and at its own cost and expense, cause the relevant Brokers to issue a written undertaking in favour of Lombard in such form as Lombard may reasonably approve having regard to market practice at such time which shall, subject as aforesaid, inter alia, (i) require such Brokers to notify Lombard forthwith upon such Brokers becoming aware of any cancellation or non-renewal of or modification to a required Insurance or of their ceasing to act as insurance brokers to the Customer or of any other fact or circumstance of which they may at any time learn which might invalidate any required insurance or render it wholly or partially unenforceable and (ii) require the Brokers to provide Lombard with a certificate of insurance at the same time as the Broker supplies the same to the Customer. The Customer shall, in relation to the Aircraft, forthwith notify Lombard of any event (including but not limited to a Total 10.9 Loss) which will or may give rise to a claim under any required Insurance. 10.10 The Customer shall not in relation to any required Insurance taken out in respect of the Aircraft: 10.10.1 make or permit the making of any modification to any required Insurance without the prior written consent of Lombard; 10.10.2 do or omit to do or permit to be done or left undone anything whereby any required Insurance would or might reasonably be expected to be rendered in whole or in part invalid or unenforceable and, without prejudice to the foregoing, shall not use or keep or permit the Aircraft or any part thereof to be used or kept for any purpose, in any manner or in any place not covered by the required Insurances; or 10.10.3 cause or knowingly permit the Aircraft to be employed or used in any place or in any manner or for any purpose inconsistent with the terms of or outside the cover provided by any required insurances or if the insurances have lapsed. 10.11 The Customer shall, in relation to any required Insurance taken out in relation to the Aircraft, bear any part of any loss or liability which is to be borne by the insured under any required Insurance (notwithstanding that such Insurance names Lombard or other person as additional named insured). 10.12 The Customer shall, in relation to any required Insurance taken out in relation to the Aircraft, reimburse Lombard on demand the amount of any premiums or premium instalments which Lombard may pay together with interest thereon calculated for the period from the date of payment by Lombard until the date of payment by the Customer (after as well as before judgment). 10.13 If the Customer shall fail to maintain or cause to be maintained Insurances which in the opinion of Lombard meet the requirements hereof, Lombard may at its option, after notice in writing to the Customer, obtain such Insurances and in such event the Customer shall, upon demand, reimburse Lombard for the cost thereof together with interest thereon for the period from the date of payment by Lombard to the date of payment by the Customer (after as well as before judgment). 10.14 Without in any way limiting the Customer's obligations under the other provisions of this clause 10, the Customer undertakes that if, at any time, clause "AVN2000 Date Recognition Exclusion Clause" or any equivalent clause is endorsed on the policies of insurance required hereunder, the Customer shall, at its cost and expense, do all things and take any and all such action available to it to ensure that the interests of Lombard under such policies of insurance are fully protected (including, without limitation, ensuring that clauses "AVN2001 Date Recognition Limited Coverage Clause" and "AVN2002 Date Recognition Limited Coverage Clause" or any equivalent clauses are endorsed on the policies of insurance required hereunder as soon as possible thereafter). If at any time Lombard shall be of the opinion (acting reasonably) that the interests of Lombard in the insurances are not adequately protected by reason of the operation of clause "AVN2000 Date Recognition Exclusion Clause" or any equivalent clause, Lombard may (but shall not be obliged to) take out such policy of insurance to insure Lombard against such risks as Lombard may think fit (acting reasonably) and the Customer shall indemnify and does hereby indemnify Lombard against any and all costs incurred in connection therewith. 11 Requisition

The Customer will remain fully liable to Lombard for all its obligations under this Deed should there be a Requisition of the Aircraft (other than those with which it is physically unable to comply owing to the requisition) unless and until

11.1

Lombard is satisfied that the requisition authority has assumed the Customer's obligations to Lomb has so confirmed to the Customer. 11.2 The Customer agrees that the Requisition Compensation will be paid to Lombard by way of Customer's Obligations if Lombard requests.	
11.2 The Customer agrees that the Requisition Compensation will be paid to Lombard by way of Customer's Obligations if Lombard requests.	f prepayment of the
11.3 When any Requisition ends the Customer will immediately and fully comply with all of its obligation	ns under this Deed.
12 Indemnity to Lombard	
12.1 The Customer will indemnify Lombard against any liability or loss (including legal costs) arising fr result of (a) the Customer or Lombard entering into this Deed ; (b) the Customer failing to perform a under this Deed ; (c) Lombard investigating any event that it believes to be a default under this De relying on any notice, request or instruction delivered under this Deed . The Customer will pay Lo Business Days of demand by Lombard .	any of its obligations
12.2 The Customer shall indemnify Lombard against any Taxation or similar charge imposed by any gov authority which arises in relation to the Aircraft , this Deed or any payments due under this Dee avoidance of doubt, any amount of value added tax determined by Lombard to be irrecoverable but Lombard's overall net profits). The Customer will pay Lombard within three Business Days of de	excluding for the excluding Taxes on
12.3 Where the Customer indemnifies or reimburses Lombard for any liability, loss or Tax the amount indemnity or reimbursement shall take account of the likely Tax treatment in the hands of Lombard determined by Lombard) of (i) the indemnity payment or reimbursement and (ii) the Tax or loss st which that amount is payable. In the event that the assumed Tax treatment of the indemnit reimbursement proves to be incorrect the Customer agrees to pay, on demand, any additional sum that Lombard occupies the same after- Tax position as it would occupy if no such adjustment were reader.	ard (as conclusively uffered in respect of ity payment or any n required to ensure
12.4 These indemnities will continue in force notwithstanding any Event of Default or termination of this	Deed.
13 When security becomes enforceable	
Without prejudice to Lombard's other rights under this Deed or by virtue of applicable law, upon the Event of Default, and at any time thereafter, Lombard may, irrespective of whether a notice has Customer, declare the security constituted by the Security Documents to have become immediate thereafter Lombard shall be entitled, without further notice or further demand, immediately to exert and remedies possessed by it according to applicable law as mortgagee of the Aircraft and the oth and in particular but, without limitation, the powers set out in this Deed.	been served on the lely enforceable and ercise all the powers
14 Possession and exercise of powers	د . ایند آند افران روهی
14.1 Lombard does not have an immediate right to possession of the Aircraft or its income (and will not taking possession if it enters to inspect or repair the Aircraft). The Customer will continue in posses takes possession.	
14.2 Any purchaser or third party dealing with Lombard or a receiver may assume that Lombard's power are exercisable without proof that demand has been made.	ers have arisen and
14.3 The receipt of Lombard or its delegate or any receiver shall be conclusive discharge to a purch disposal of any of the Security Assets .	naser in any sale or
15 15 Appointment of receiver	en Antonio (1990) Antonio (1990) Antonio (1990)
Following the occurrence of an Event of Default, Lombard may appoint or remove a receiver or receiver Any restriction imposed by law on the right of a mortgagee to appoint a receiver (including under s Law of Property Act 1925) that can be excluded does not apply to this Deed . If Lombard appoints a may agree and pay the receiver's fees and expenses (without being limited to the maximum rate 109(6) of the Law of Property Act 1925). The receiver will be the Customer's agent and the Lombard) will be responsible for the acts, defaults and remuneration of the receiver.	section 109(1) of the a receiver, Lombard specified in section
16 Powers of Lombard and receivers	د. بر ز در معاد رومه بر رومه کرد.
16.1 When the security created by this Deed becomes enforceable in accordance with clause 13, Lomb may:	bard or any receiver
16.1.1 enter, take possession of, operate and/or generally manage the Aircraft and enter upon any land or Aircraft may be located without being liable to the Customer (or any lessee of the Aircraft) for o entry;	premises where the or by reason of such

apply to any authority in the country in which the Aircraft or any part thereof is located for an enforcement and attachment 16.1.2 order in respect of the Aircraft; 16.1.3 apply to any court of competent authority for an order so as to vest all the Customer's right, title and interest in the Aircraft in Lombard absolutely; 16.1.4 - sell, call in, collect and convert into money the Aircraft to the extent that it is entitled to do so with all such powers in that respect as are conferred by applicable law (the power of sale shall arise, and the Customer's Obligations shall be deemed due and payable for that purpose, on the execution of this Deed) and by way of extension thereof such sale, calling in, collection and conversion may be made by public or private contract at any place in the world with or without notice to the Customer, without being liable to account for any loss of or deficiency in the consideration therefore and so that Lombard shall have power in its absolute discretion to postpone any such sale or to proceed to a forced sale without being liable for any loss resulting there from (except that resulting from Lombard's wilful misconduct or gross negligence). and for the purposes of this clause 16.1.4 Lombard may (in its sole discretion) by written notice appoint the Customer its non-exclusive agent to sell the Aircraft on terms satisfactory to Lombard; 16:1.5 maintain and/or repair the Aircraft; 16.1.6 settle, arrange, compromise or submit to arbitration any accounts, claims, questions or disputes whatsoever which may arise in connection with or in any way relating to the Aircraft or the Customer's Obligations and execute releases or other discharges in relation thereto; put the Aircraft in storage for such term and generally in such manner and upon such conditions and stipulations as 16.1.7 Lombard shall think fit; 16.1.8 execute and do all such acts, deeds and things as Lombard may consider necessary or proper for or in relation to any of the purposes aforesaid; 16.1.9 hire, lease, charter or accept surrenders of any hire, lease or charter, charge or deal with the Aircraft without restriction; 16.1.10 take, continue or defend any proceedings and enter into any arrangement or compromise in relation to the Security Assets; 16.1.11 insure the Aircraft, arrange indemnity and other similar insurance, and obtain bonds and give counter-indemnities and other security in connection with this Deed; 16.1.12 employ advisers, consultants, managers, agents and others; 16.1.13 purchase or acquire materials, tools, equipment, furnishing, goods or supplies; 16.1.14 collect, recover and give good discharge for any moneys or claims forming part of, or relating to, any Security Assets and to permit any brokers through whom collection or recovery is effected to charge the usual brokerage for such collection or recover; and 16.1.15 do any acts which Lombard or a receiver considers to be incidental to the exercise of its powers or generally beneficial to Lombard. 16.2 A receiver may borrow and secure the repayment of any money, in priority to the Customer's Obligations. 16.3 In addition to the powers set out in this clause 16, a receiver shall have all the powers conferred on a receiver by the Law of Property Act 1925 and the Insolvency Act 1986 and any other applicable statute. Sections 93 and 103 of the Law of Property Act 1925 shall not apply to this Deed. 16.4 Joint receivers may exercise their powers jointly or separately. A receiver will first apply any money received from the Aircraft towards the repayment of all money that the receiver has 16.5 borrowed and secondly in payment of the receiver's fees and expenses. The receiver will then apply any remaining money received as required by law. Lombard may exercise any of its powers whether or not a receiver has been appointed. 16.6 Each of Lombard and any receiver shall have full power to delegate (either generally or specifically) the powers, 16.7 authorities and discretions conferred on it by this Deed on such terms and conditions as it sees fit, which delegation shall not preclude either the subsequent exercise of such power, authority or discretion by Lombard or the receiver itself or any subsequent delegation or revocation of it.

16.	.8	Any credit balance with Lombard will not be repayable, or capable of being transferred, charged or dealt with by the Customer until the Customer's Obligations have been paid in full.
16	.9	None of Lombard , its delegate(s), nominee(s) or any receiver shall be liable for any loss by reason of (a) taking any action permitted by this Deed or (b) any neglect or default in connection with the Security Assets or (c) taking possession of or realising all or any part of the Security Assets , except in the case of gross negligence or wilful default upon its part.
16.	.10	Lombard will not be required in any manner to perform or fulfil any obligation of the Customer, make any payment, make any enquiry as to the nature or sufficiency of any payment received by it or present or file any claim or take any action to collect or enforce the payment of any amount.
16.	.11	Lombard shall not be liable either to the Customer or to any other person by reason of the appointment of a receiver or for any other reason.
16.	.12	Neither Lombard nor the receiver will be in any way liable or responsible to the Customer for any loss or liability arising from any act, default, omission or misconduct on the part of any delegate or sub-delegate.
17		Application of payments
17.	.1	Lombard may apply any payments received for the Customer to reduce any of the Customer's Obligations, as Lombard decides.
17.	.2	Until all amounts which may be or become payable by the Customer to Lombard have been irrevocably paid in full, Lombard (or any trustee or agent on its behalf) may without affecting the liability of the Customer under this Deed:
17.	.2.1	refrain from applying or enforcing any other monies, security or rights held or received by Lombard (or any trustee or agent on its behalf) against those accounts or apply and enforce them in such manner and order as it sees fit (whether against those amounts or otherwise); and
17.	.2.2	hold in an Interest-bearing suspense account any moneys received from the Customer or on account of the Customer's liability under this Deed.
17.	3	If Lombard reasonably considers that any amount paid or credited to it is capable of being avoided or reduced by virtue of any bankruptcy, insolvency, liquidation or similar laws, the liability of the Customer under this Deed and the security created under this Deed shall continue and such amount shall not be considered to have been irrevocably paid.
17.	.4	If Lombard receives notice of any Encumbrance affecting the Aircraft other than a Permitted Lien, Lombard may suspend the operation of the Customer's account(s) and open a new account or accounts. Regardless of whether Lombard suspends the account(s), any payments received by Lombard after the date of that notice will be applied first to repay any of the Customer's Obligations arising after that date.
18		Preservation of other security and rights and further assurance
18.	.1	This Deed is in addition to any other security for the Customer's Obligations held by Lombard now or in the future. Lombard may consolidate this Deed with any other security so that they have to be redeemed together, but it will not merge with or prejudice any other security or Lombard's other rights.
18.	.2	The security created under this Deed will be a continuing security for the ultimate balance of the Customer's Obligations notwithstanding any intermediate payment or settlement of all or any part of the Customer's Obligations .
18.	.3	The obligations of the Customer under this Deed will not be affected by any act, omission or thing which, but for this provision, would reduce, release or prejudice any of its obligations under this Deed (whether or not known to it or Lombard). This includes:
18.	.3.1	any time or waiver granted to, or composition with, any person;
18.	3.2	any release of any person under the terms of any composition or arrangement;
18.	.3.3	the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce any rights against, or security over assets of, any person;
18.	3.4	any non-presentation, non-observance of any formality or other requirements in respect of any instrument or any failure to realise the full value of any security;
18.	3.5	any incapacity, lack of power, authority or legal personality of or dissolution or change in the members or status of any person;
18.	3.6	any amendment of any document or security;

18.3.7	any unenforceability, illegality, invalidity or non-provability of any obligation of any person under any document or security or the failure by any person to enter into or be bound by any document or security; and
18.3.8	any insolvency or similar proceedings.
18.4	Without prejudice to the generality of clause 18.2 above, the Customer expressly confirms that it intends that the security under this Deed shall extend from time to time to any variation, increase, extension or addition (however fundamental) of or to any document creating any of the Customer's Obligations and/or any facility or amount made available under such document.
18.5	Unless all amounts which may be or become payable by the Customer to Lombard have been irrevocably paid in full or as Lombard otherwise directs, the Customer will not, after a claim has been made or by virtue of any payment or performance by it under this Deed :
18.5.1	be subrogated to any rights, security or moneys held, received or receivable by Lombard (or any trustee or agent on its behalf); nor
18.5.2	be entitled to any right of contribution or indemnity in respect of any payment made or moneys received on account of the Customer's liability under this clause.
18.6	The Customer waives any right it may have of first requiring Lombard (or any trustee or agent on Lombard's behalf) to proceed against or enforce any other right or security or claim payment from any person or file any proof or claim in any insolvency, administration, winding-up or liquidation proceedings relative to any other person before claiming from the Customer under this Deed.
18.7	If the Customer's Jurisdiction of Incorporation, or the State of Registration has, or at any time brings into force, any legislative or other provisions giving effect to the Cape Town Convention and Aviation Protocol, the Customer shall:
18.7.1	promptly register together with Lombard any interest with the International Registry with respect to the Aircraft and/or any Engine and constituted by any Finance Document;
18.7.2	promptly register together with Lombard any interest with the International Registry with respect to any replacement Engine provided in accordance with clause 8.1.3 at the time of such replacement;
18.7.3	promptly do all acts and things and enter into any agreements (subordination or otherwise) reasonably required by Lombard to protect and/or enhance and/or improve the title, rights, priority of any international interests and any other interests vested in Lombard with respect to the Aircraft and/or any Engine and constituted by any Finance Document and consent to the registration of any of the foregoing;
18.7.4	agree to any amendment, restatement, revision or any variation to any Finance Document to the extent necessary if required by Lombard to permit the interests created by the relevant Finance Document to constitute international interests under the Cape Town Convention and Aircraft Protocol;
18.7.5	promptly execute and deliver an irrevocable de-registration and export request authorisation in respect of the Aircraft in the form set out Schedule 6 (an "IDERA");
18.7.6	use it best endeavours to promptly procure countersignature of the IDERA by the Aviation Authority and return an original IDERA signed by the Customer and the Aviation Authority to Lombard; and
18.7.7	not register or consent to the registration of, any conflicting interests (whether or not taking priority over Lombard's international interests) at the International Registry without Lombard's prior written consent.
18.8	The Customer shall not change the State of Registration without the consent of Lombard (which Lombard shall have absolute discretion to withhold). At the request of Lombard upon a change of the State of Registration of the Aircraft taking place, the Customer shall execute in favour of Lombard a deed amending or supplementing this Deed and/or a mortgage and/or any other document required by Lombard for the purpose of preserving the security intended to be created hereby in such form as Lombard may reasonably request.
19	Non-compliance by the Customer
	If the Customer for any reason fails to observe or punctually perform any of its obligations under this Deed , Lombard shall be entitled, on behalf of or in the name of the Customer or otherwise and at the cost of the Customer , to perform the obligation and to take any steps which Lombard may consider necessary with a view to remedying or mitigating the consequences of the failure.
20	Power of attorney
n de la composition de la composition de la composition	a de la seconda de la companya de la seconda de la sec 1914 - La seconda de la sec 1914 - La seconda de la sec

	By way of security, to give effect to this Deed and to perfect, protect and enhance the security created by this Deed and secure the exercise of any of their powers, the Customer irrevocably appoints Lombard , and separately any receiver appointed pursuant to the provisions of this Deed , to be the Customer's attorney (with full power of substitution and delegation), in the Customer's name to sign or execute any documents, deeds and other instruments, or take, continue or defend any proceedings in accordance with this Deed and the Aviation Loan Agreement . The Customer ratifies and confirms whatever any attorney does or purports to do pursuant to its appointment under this clause 20.
21	Lombard's rights
	If Lombard waives or decides not to enforce any of its rights under this Deed, Lombard may enforce any such right at a later date.
22	n 1. Set-off – Larren en e
	Lombard may at any time, and without notice to the Customer, combine or consolidate all or any of the Customer's then existing accounts with any liabilities to Lombard and apply any sum due to the Customer in satisfaction of any sum due and unpaid by the Customer under any Finance Document or any other agreement, or guarantee with or other obligation owed to Lombard.
23	Currency
23.1	Lombard may convert any payment it receives in any other currency in respect of this Deed to the Currency at its chosen spot rate of exchange for the purchase of the relevant currency in the London foreign exchange market at the relevant time.
23.2	Lombard may convert the Value stated in any valuation supplied to it in connection with this Deed from one currency to the Currency, at its chosen spot rate of exchange for the purchase of the relevant currency in the London foreign exchange market at the relevant time.
24	Severability
	If at any time any provision in connection with this Deed becomes illegal, invalid or unenforceable the remaining provisions will not in any way be affected or impaired thereby.
25	· Variations
	No variation of these terms shall be valid unless it is in writing and signed by or on behalf of each of the Customer and Lombard.
26	n 1. Notices 1. Notices
26.1	All consents, notices and demands relating to this Deed must be in writing (which shall include by electronic means).
26.2	Lombard may deliver a notice or demand to the Customer at the contact details last known to Lombard, the Customer's registered office, or the Address for Service, if one is specified.
26.3	A notice or demand by Lombard will be deemed given at the time of personal delivery; on the next Business Day after posting; or, if by fax or other electronic means, at the time of sending, if sent before 6.00 pm (London time) on a Business Day , or otherwise on the next Business Day .
26.4	The Customer must deliver a notice to Lombard addressed to the address or email address shown above. Notice by the Customer will be effective when received by Lombard, provided that any email received on any day other than a Business Day, or at any time after 6.00pm (London time) on a Business Day, shall not be effective until 9.30am (London time) on the next Business Day.
27	n 1. Transfers 1. Transfers
27.1	Lombard may allow any person to take over any of its rights or duties under this Deed and may provide information about the Customer or any Finance Document to which the Customer is party to the transferee in connection with any proposed transfer and any person to whom such information is required to be disclosed by any applicable law.
27.2	The Customer is not entitled to assign or otherwise transfer its rights or obligations under this Deed.
28	Counterparts
1 60	

	77 <u>121222222222222222222222222222222222</u>
) Third party rights	
Unless expressly provided to the co Contracts (Rights of Third Parties) A	ontrary in this Deed , a person who is not a party to this Deed has no right under the Act 1999 to enforce or to enjoy the benefit of any term of this Deed .
Entire agreement	
	ns made in accordance with the terms of this Deed) ,and any documents referred t ent between the parties. It supersedes any previous agreement between them i Deed .
Customer Information	
Appendix 1 details who Lombard m	nay share Customer information with and for what purpose(s).
2 Law	
This Deed and any non-contractual	obligations arising out of or in connection with it are governed by English law.
and binding on the Customer and r extent allowed by law, Lombard ma	or ruling in any proceedings connected with the Deed in those courts will be conclusive may be enforced against the Customer in the courts of any other jurisdiction. To the ay take concurrent proceedings in any number of jurisdictions. ied, or any other address provided for this purpose, will be an effective address for
serving proceedings on the Custon	
hem on the date stated at the beginning of	uis veeu.
Executed and Delivered as a deed by TAC	
u u	
Executed and Delivered as a deed by TAC	G Aviation (Stansted) Ltd
Executed and Delivered as a deed by TAC	G Aviation (Stansted) Ltd Director/Secretary
Executed and Delivered as a deed by TAC Director Print name	G Aviation (Stansted) Ltd Director/Secretary
Executed and Delivered as a deed by TAC Director Print name f only one signature is required, it must be	G Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name If only one signature is required, it must be Signed by the Director in the presence of:	S Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name If only one signature is required, it must be Signed by the Director in the presence of: Witness' signature	C Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name If only one signature is required, it must be Signed by the Director in the presence of: Witness' signature	S Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name f only one signature is required, it must be Signed by the Director in the presence of: Witness' signature	S Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name f only one signature is required, it must be Signed by the Director in the presence of: Witness' signature Witness' name in full Address Doccupation	S Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name f only one signature is required, it must be Signed by the Director in the presence of: Witness' signature Witness' name in full Address Doccupation Signed for by Lombard North Central PLC Authorised Signatory	S Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name If only one signature is required, it must be Signed by the Director in the presence of: Witness' signature Witness' name in full	S Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required
Executed and Delivered as a deed by TAC Director Print name If only one signature is required, it must be Signed by the Director in the presence of: Witness' signature Witness' name in full Address Doccupation Signed for by Lombard North Central PLO Authorised Signatory Z&HWNT	S Aviation (Stansted) Ltd Director/Secretary Print name that of a Director, and a witness is required